IWQW

Institut für Wirtschaftspolitik und Quantitative Wirtschaftsforschung

Diskussionspapier Discussion Papers

No. 06/2015

WEIGHT LOSS AND SEXUAL ACTIVITY IN ADULT OBESE INDIVIDUALS: ESTABLISHING A CAUSAL LINK

Arndt R. Reichert The World Bank and RWI

Harald Tauchmann University of Erlangen-Nürnberg, RWI and CINCH

> Ansgar Wübker RWI

ISSN 1867-6707

WEIGHT LOSS AND SEXUAL ACTIVITY IN ADULT OBESE INDIVIDUALS: ESTABLISHING A CAUSAL LINK*⁺

Arndt R. Reichert

The World Bank Rheinisch-Westfälisches Institut für Wirtschaftsforschung (RWI)

Harald Tauchmann

University of Erlangen-Nuremberg Rheinisch-Westfälisches Institut für Wirtschaftsforschung (RWI) CINCH (Health Economics Research Centre)

Ansgar Wübker

Rheinisch-Westfälisches Institut für Wirtschaftsforschung (RWI)

June 19, 2015

Abstract

Obesity may not only be linked to undesirable health outcomes but also to limitations in sexual life. The present paper aims to assess whether there is a causal relationship between weight loss and sexual activity in adult obese individuals. To address the endogeneity of weight loss that is likely to result in biased estimation results, the analysis is based on data from a randomized field experiment. In this experiment financial weight-loss rewards were offered to a random subgroup of participants and can be used as exogenous source of weight variation in an instrumental variables approach. Estimation results indicate that for obese males loosing weight increases the probability for being involved in a sexual relationship. Conditional on having already lost some weight, a further reduction in obesity also increases the frequency of sexual intercourse.

JEL codes: 110, 118, J28, J65 *Keywords:* obesity, sexual partnership, frequency of intercourse, randomized trial, weight-loss incentives.

^{*}The authors are grateful for valuable comments to Nils Heinrich, Christiane Wuckel, Simon Reif, and to the participants of the Economics Research Seminar at the University of Mainz.

⁺Address for correspondence: Harald Tauchmann, Professur für Gesundheitsökonomie, Findelgasse 7/9, 90402 Nürnberg, Germany. Email: harald.tauchmann@fau.de.

1 Introduction

A close link of a fulfilling sex life and general life satisfaction is well documented in the literature (e.g. Woloski-Wruble et al., 2010; Schafer et al., 2013). Impairments in sexual life thus are likely to have a strong negative impact on individual happiness and life satisfaction. One possible reason for limitations in sexual life is obesity; see Larsen et al. (2007) for a review of the medical literature focussing on the association of obesity and sexual dysfunction such as erectile dysfunction. Though the majority of analyses reviewed by Larsen et al. (2007) find a positive association, the evidence is still mixed.¹ Taking a less physiological oriented perspective and considering deficits in sexual life beyond sexual dysfunctions, Kolotkin et al. (2006) list 'lack of enjoyment', 'lack of desire', 'difficulties with performance', and 'avoidance of sexual encounters' as examples for obesity related impairments of sexual life. Thus, besides undesirable general health outcomes, obesity may have a more direct detrimental effect on happiness and life satisfaction that operates through an unsatisfactory sex life.

The non-medical, health-economics oriented literature has taken a very different view on the link of body weight to sexual life. This relatively small literature (Cawley, 2001; Halpern et al., 2005; Cawley et al., 2006; Sabia and Rees, 2011; Ali et al., 2014; Neymotin and Downing-Matibag, 2014) focusses on individuals of very young age, adolescents in particular. Due to the focus on this special age group, sexual activity is not primarily interpreted as one facet of a happy and fruitful life. It is rather associated with premature initiation to sex, potentially exerting detrimental effects on adolescents' later lives. For female adolescents Sabia and Rees (2008) find a causal and detrimental effect of early sexual intercourse on psychological well-being. Ali et al. (2014) allude to disease and pregnancy risks teenagers take when having sex early in their lives, suggesting that a postponed initiation to sexual activities might be a beneficial side-effect of teenage overweight.

The contribution of the present paper is to (i) conduct an analysis that focusses on adult individuals, for whom the above line of argument does not apply and whose sexual behavior, most likely, differs from the behavior of the age group studied by the above cited literature. Moreover, unlike the majority of the medical literature, our focus is (ii) on (self-reported) sexual activity rather than specific sexual dysfunctions such as erectile dysfunction. This is important as obesity related limitations in sex life may often be unrelated to any physiological deficit but can possibly be attributed to social and psychological body-weight related factors, such as being less attractive

¹Especially for women, for which 'sexual dysfunction' is less clearly defined than for men and for which the number of studies is much smaller, several analyses fail in establishing an association of sexual dysfunction and body weight (e.g. Adolfsson et al., 2004; Kolotkin et al., 2006).

to a potential partner, a lack of self confidence when dating, or less enjoyment in and desire for sex. Finally, the present analysis contributes to the existing evidence by (iii) establishing a causal effect of weight-loss on sexual activity as opposed to finding a mere correlation in the data.

Focussing on the non-medical strand of the literature, causality in the link between body weight and sex life has not always attracted much attention. Halpern et al. (2005), for instance, estimate that one BMI unit less increases the probability of being involved in a romantic relationship (with or without sexual intercourse) by 6-7 percent in adolescent girls. Using Add Health (National Longitudinal Survey of Adolescent Health; Harris, 2013) data,² a US longitudinal school-based survey, Cawley et al. (2006) find that for obese girls the odds of initiation to sexual intercourse are just 32 percent of the odds of normal-weight girls. However, both analyses do not account for unobserved heterogeneity. Hence, the striking results may well be attributed to unobserved confounding factors, such as personal character traits, that are relevant for both obesity related behavior (eating habits, physical activity) and mating behavior. Numerous studies from the medical literature (see Larsen et al., 2007) likewise are concerned with the association of overweight and impairments in sex life rather than with a causal relationship.

More recent papers address possible endogeneity by the use of instrumental variables (IV) estimation. Also analyzing data from Add Health and using siblings' BMI and mother's obesity status as instruments for the respondents' body weight, Sabia and Rees (2011) find that a one unit increase in BMI reduces the probability ob being sexually active by 3.5 percentage points in girls aged 14 to 17. In contrast, no significant effect is found for boys. Rather, the point estimate for boys bares the opposite sign in several specifications. Ali et al. (2014) take a similar approach by instrumenting body weight with maternal obesity status. For white girls – but not for blacks – they find that higher body weight or being obese significantly lowers the probability of having had sex or having been involved in a romantic relationship.

Though both papers carefully discuss the validity of the instruments used and argue that including a wide range of covariates does capture any possible direct effects from mother's body weight to the child's sexual behavior, one may still doubt whether mother's – and other relatives' – body weight is a valid instrument. One concern is that not only body weight but also sexual behavior – or at least attitudes towards it – may be intergenerationally transmitted from the mother to her children (e.g. Taris, 2000). If so, mother's own sexual behavior represents another channel through which maternal body weight influences children's sex life, given that body weight matters for sexual behavior. Another concern with respect to instrument validity is the social en-

²When alternatively using the NLSY (National Longitudinal Survey of Youth) Cawley et al. (2006) cannot confirm their key Add Health based findings.

vironment children share with their mothers and siblings. It may strongly influence both body weight and attitudes towards sex for both mothers and children and, most likely, cannot entirely be controlled for by including covariates in the analysis.

While following Sabia and Rees (2011) and Ali et al. (2014) in relying on instrumental variables for establishing a causal link between body-weight and sexual behavior, we do not use maternal body weight as instrument. Rather, in our analysis identification rests on exogenous variation in BMI that was artificially induced in a randomized field experiment. More precisely, we use data that originate from a randomized trial that was conducted to evaluate the effectiveness of financial incentives for weight loss (Augurzky et al., 2012, 2014). In this experiment individuals in the intervention groups were exposed to financial incentive for loosing body weight, while control group members were not. Hence following Reichert (2015) treatment status can serve as an instrument for the change in body weight. Besides using different instruments and focussing on adult individuals, our analysis materially deviates from Sabia and Rees (2011) and Ali et al. (2014) by only considering obese individuals. That is, the empirical analysis is not concerned with the question of whether being overweight, or even obese,³ has adverse effects on sex life. Rather, we are interested in the question of whether – even moderate – weight loss in obese individuals makes a difference for their sexual lives.

The strategy for identification of the present analysis is related to the approach of Esposito et al. (2004). This medical study also relies on data from a randomized field experiment with obese participants. In this study the intervention was not exposition to financial incentives but guidance on how to loose weight combined with weight monitoring through monthly group sessions. Yet, rather than identifying a causal effect of weight loss, Esposito et al. (2004) estimate a reduced form effect by comparing post-intervention IIEF (International Index of Erectile Function) scores between the intervention and the control group. The statistically significant advantage in IIEF found for the intervention group nevertheless provides striking evidence for body-weight mattering for sexual dysfunctions in obese. However, by focussing on erectile dysfunction alone, Esposito et al. (2004) may miss out other obesity related impairments of sexual life. By considering two different measures of sexual activity our analysis takes a broader perspective on the link between obesity and sex life.

The remainder of this paper is organized as follows. The subsequent section 2 introduces the data and describes the experiment the data originates from. Section 3 discusses the econometric model and section 4 presents the estimation results for the basic model. In section 5 alternative

³The WHO defines overweight and obesity as the body mass index (BMI, body weight per squared body height $[kg/m^2]$) reaching the threshold values of 25 and 30, respectively (World Health Organization, 2000).

model specifications and estimation results are discussed. Section 6 summarizes and discusses the main findings and concludes.

2 Data

2.1 The Experiment

The data used in this analysis originate from a large scale, four-phase, randomized field experiment that is described in more detail elsewhere (Augurzky et al., 2012). The experiment was conducted between March 2010 and July 2013. Roughly 700 obese individuals where recruited in the period March 2010 to August 2011. Recruitment took place in four rehabilitation clinics in the south-west of Germany (Bad Mergentheim, Bad Kissingen, Isny, and Glottertal). Only for a minor group of participants obesity was the immediate reason for their rehab stay. Yet, many were sent to the clinic because of health deficits that are connected with overweight or obesity such as chronical back pain, for instance. The participating rehab clinics are operated by the German Pension Insurance (DRV), Baden-Württemberg section. For this reason all patients are insured with DRV and the ultimate aim of treatment at the rehab clinics is either to preserve or to restore patients' workableness. Shortly after clinic admission the physician in charge approached all patients who had a BMI of 30 or above and complied with the further admission criteria⁴ and asked them for voluntary participation in the experiment. Shortly before discharge from the clinic, after a typically three weeks stay, all participating individuals were set an individual weight-loss target by the physician in charge that they were prompted to realize within four months. The physicians were advised to choose a weight-reduction target of about six to eight percent of current body weight, but where free in setting a target they regarded medically appropriate given the specific case. Subsequently the participants were randomly assigned to two incentive groups (*incentive* 150_{1-4} and *incentive* 300_{1-4})⁵ and one control group (*incentive* 0_{1-4}). Members of the former two groups were promised \in 150 and \in 300, respectively, for realizing (or exceeding) their individual weight-loss target within four months, while members of the control group were not exposed to any financial incentives for reducing body weight. Members of *incentive* 150_{1-4} and *in*centive 3001-4 who fail in fully complying with the scheduled weight loss but managed to reduce

⁴Age between 18 and 75 years, resident of the federal state of Baden-Württemberg, sufficient German language skills, no pregnancy, no psychological and eating disorders, no substance abuse, no seriously illnesses (specified list of diseases).

⁵To keep notation simple we omit the index *i* indicating individuals. Numerical subscripts refer to months since rehab discharge. Thus, variables indexed with a simple numerical subscript refer to a certain point in time; e.g. BMI_0 denotes BMI measured at rehab discharge while BMI_4 denotes BMI measured four months later, i.e. by the end of the weight-loss phase. Variables indexed with a bipartite numerical subscript separated by a short hyphen refer to a time span; e.g. for *incentive* 300₁₋₄ the observation period is month 1 to month 4 (including months 1 and 4), i.e. the entire four months weight-reduction phase.

their body weight by at least 50 percent of the contractual target were awarded proportionally to the degree of target achievement. That is a member of *incentive* 300_{1-4} with body weight of 100 kg at clinic discharge and a weight-loss target of 7 kg would not have received any financial rewards if she were weighed more than 96.5 kg at the end of the weight-reduction phase. She would have received the full premium of \in 300 if they were weighed 93 kg or less. If she were weighed *w* kg, with 93 < $w \le$ 96.5, she would have received \in 300 · (100–*w*)/7.

The weigh-in at the end of the weight-reduction phase – and all further weigh-ins – were carried out at assigned pharmacies. While a pharmacy as close as possible to the participants place of residence was typically selected by the experimenters, the test persons could not actively influence to which particular pharmacy they were sent.⁶ Thus, the location of weigh-in was exogenous to the participants. All participants received a show-up fee of €25 for attending the weigh-in and (partially) successful incentive group members received the respective reward. Shortly after the weigh-in, all participants – irrespective of group membership in the weight-reduction phase and irrespective of success - were prompted to comply with their target weight by the end of a subsequent six months weight-maintenance phase.⁷ Yet, conditional on success in the weight reduction phase, a second randomization took place at the same time. Participants who had lost at least 50 percent of their contractual weight-loss target were randomly assigned to two intervention groups (*incentive* 250_{5-10} , *incentive* 500_{5-10}) and one control group (*incentive* 0_{5-10}). The former two could gain up to $\in 250$ and $\in 500$, respectively, if they were weighed their target weight or less ten months after rehab discharge. For those who did not fully comply with the target weight, the actual reward was calculated using the same rule as described above. No rewards were promised to members of the control group. Individuals who were not successful in the weight-reduction phase were hence effectively assigned to the control group.

Six month later, i.e. ten months after rehab discharge, another weigh-in took place under identical conditions. Yet, at this point in time the incentive scheme was terminated and no further randomization took place. However, all participants were still requested to comply with their target weights and another weigh-in was announced to take place after another twelve months. This final year serves the follow-up phase. See Figure 1 for the time line of the entire experiment.

By the end of each of the four phases, the participants had to answer a detailed questionnaire. The final one, which was sent to the participants 22 months after rehab discharge, is of major importance to the present analysis as it is the only one that includes questions concerning the participants sex life. Yet, only for participants with even identification number – i.e. for a fifty

⁶Any pharmacy was contacted by the experimenter prior to sending participants there for the weigh-in. If a pharmacy denied cooperation – though it was remunerated for its effort – a more distant pharmacy had to be assigned.

⁷For not fully successful individuals the 'weight-maintenance phase' was effectively another weight-reduction phase.



Figure 1: Time line of the four-phase experiment. *Source:* Augurzky et al. (2014).

percent random sample⁸ – the questionnaire included these delicate questions. The reason for this was to reduce the risk of making too many individuals drop out by asking them – possibly displeasing – questions about their sex lives. However, this concern proved immaterial as the drop out rate virtually did not differ between individuals with even and with odd identification number.

The experiment population was subject to significant sample attrition. 697 individuals started the weight-reduction phase after rehab discharge. 177 participants dropped out during the weight-reduction phase, i.e. they did not show up at the weigh-in after four months, leading to only 520 individuals who entered the weight-maintenance phase. In this phase another 109 individuals dropped out. The follow-up phase was also subject to some sample attrition reducing the number of individuals for whom weight information is available by the end of the experiment to 316. Among these 174 were asked questions regarding sex life. Only 17 denied any kind of information on their sex lives. This corresponds to a rate of item-non-response of less than 10 percent, which is remarkably low compared to many surveys addressing sexual behavior (cf. Fenton et al., 2001).

2.2 Variables

In the final questionnaire, which participants had to fill in 22 months after rehab discharge, they were asked two questions regarding their sex lives, (i) whether being involved in a sexual relationship and (ii) how frequently they have sexual intercourse. Both questions refer to the previous twelve month that is to the follow-up phase of the experiment. While the fist question had to be

⁸As the individual identifiers were issued as clinic specific sequential numbers, for each participant the odds for an even or an odd id-number were fifty-fifty.

	frequency of sexual intercourse					ma	rginal distrik	oution ⁺
	never	occasionally	monthly	weekly	daily	all	males	females
sexual relationship: no yes	32 0	9 25	1 32	$\begin{array}{c} 0\\ 48 \end{array}$	$\begin{array}{c c} 0 \\ 4 \end{array}$	45 111	20 83	25 28
marginal distribution ⁺ : all males females	32 11 21	35 26 9	33 22 11	48 37 11	4 3 1	157 99 53	103	53

Table 1: Joint and Marginal Distribution of Dependent Variables

Notes: ⁺Due to item non response, values do not exactly sum up. 157 responses to (at least) one of the two questions regarding sex life; 156 responses to question about sexual relationship; 152 responses to question about frequency of intercourse.

answered by yes or no, the latter allowed for answers on a five categories scale, more precisely 'never', 'occasionally', 'at least once a month', 'at least once a week', '(almost) daily'. Hence we can use two different measures of sexual activity as dependent variables, a binary indicator (sexpartner₁₁₋₂₂) and an ordered categorial one (sexfrequency₁₁₋₂₂). 156 and 152, respectively, participants answered these questions.⁹ For model estimation, we reduced the number of categories for the latter variable to three ('never', 'occasionally', 'regularly') where the third category combines having sex monthly, weekly or (almost) daily. The reason for this is the relatively small sample of observations that - for some model specifications - renders the number of individuals that are observed in some of the original categories very small. Table 1 displays the joint distribution of sexpartner₁₁₋₂₂ and sexfrequency₁₁₋₂₂ (full set of categories). The information regarding the two measures seems to be mutually consistent, as no respondent who reports not to have had sex in the previous year, reports having been involved in a sexual partnership for the same period. The data also indicates that the respondents distinguish between living together with a partner and having a sexual relationship. Though both variables are highly positively correlated, roughly one in ten of those who live with a partner report not to live in a sexual partnership. None of these individuals reports having sex more often than occasionally.

The dependent variables are observed for the final phase of the experiment (month 11 to month 22). We measure the key explanatory variables prior to this and use the change in BMI measured between rehab discharge and the end of the weight maintenance phase, denoted ΔBMI_{1-10} , as key explanatory variable.¹⁰ For an alternative specification of the model, we use weight change over the entire experiment duration (ΔBMI_{1-22}) as explanatory variable. Thus, in this specification we allow for the measuring periods for the dependent variable and the key regressor to overlap; see Tables 13 and 14 in the Appendix for results which do not differ much from those for the preferred specification.

⁹While a single individual answered the question concerning the frequency of sexual intercourse but provided no information on involvement in a sexual relationship, for five individuals the provision of information exhibits the reverse pattern.

¹⁰One may think of using the level of the BMI (at the end of the maintenance phase) instead of its change. Yet, as long as BMI at rehab discharge is controlled for, both model variants are fully equivalent.

The small number of observations precludes specifying a rich regression model with numerous controls. Hence only age and sex¹¹ as basic socioeconomic characteristics enter the model. Besides these, BMI at rehab discharge (BMI_0) and an indicator for living together with a partner at the time of rehab discharge ($together_0$) enter the regression model. We include the former to capture the effect of pre-intervention body weight which is most likely endogenous and for which no instruments are available.¹² The latter we include as the closest substitute available for pre-intervention values of the dependent variable, which is not observed in the data. We also estimated specifications with age squared and body height – which might matter for sexual attractiveness – entering as additional controls. Both proofed statically insignificant and including them has very little effect on the estimation results. So we stick to a more parsimonious specification. Table 2 displays key descriptives for the explanatory variables.

The key concern to the present analysis is the possible endogeneity of both body weight and its change over time. The validity of estimates obtained from naively regressing measures of sexual activity on body weight are subject to concern of endogeneity bias for several reasons. First, unobserved individual characteristics may have an effect on either variable. One may hypothesise that individuals with high self-esteem are more successful in finding a partner and are also less vulnerable to overeating and obesity. This argument also applies if a change in body weight rather than its level is considered. Self-confidence and self-esteem are possibly decisive for being successful, when seriously trying to get less obese. Yet, a different channel through which individual heterogeneity may generate a spurious correlation is the individual preference for physical pleasures. If such preferences are strong, one is likely to have more desire for both having sex and calorie intake. Second, reverse causality might also be a source of endogeneity bias. One such channel is sexual frustration-induced overeating. Yet, one may also argue that being satisfied with his or her sex life reduces the pressure to comply with ideals of beauty, rendering obesity less costly in psychological terms.

As we observe the dependent variables only once, fixed effects estimation is not an option for eliminating unobserved heterogeneity. Moreover, the threat of bias due to reverse causality is not fully eliminated via regressing *sexpartner*₁₁₋₂₂ and *sexfrequency*₁₁₋₂₂ on a lagged measure of body weight (ΔBMI_{1-10}). While using a lagged regressor evidently rules out direct reverse causality, lagged body weight may still be influenced by past sex life. And, provided that sexual behavior exhibits some kind of persistency over time, unobserved past sex life is directly linked to the

¹¹For six individuals, who did not state their gender in a questionnaire, sex is imputed on basis of individuals characteristics such as body height and labor market status. Excluding these few individuals from the estimation sample, has hardly any effect on the estimation results.

¹²For this reason, the coefficient of BMI_0 is most likely estimated with bias. Yet, as our instruments for ΔBMI_{1-10} are orthogonal to BMI_0 , the bias does not carry over to the coefficient of primary interest.

	Mean	S.D.	Median	Min.	Max.
			A11		
dependent variables:			All		
sexpartner ₁₁₋₂₂	0.712	0.455	1.000	0.000	1.000
sexfrequency ₁₁₋₂₂	1.349	0.808	2.000	0.000	2.000
$together_{22}^+$	0.637	0.482	1.000	0.000	1.000
explanatory variables:					
ΔBMI_{1-10}	-1.229	2.596	-1.130	-15.347	3.774
ΔBMI_{1-22}^+	-0.318	3.305	-0.080	-18.114	10.040
ΔBMI_{5-10}^+	0.448	1.604	0.362	-8.424	4.181
age	50.166	7.982	51.000	21.000	68.000
female	0.338	0.474	0.000	0.000	1.000
together ₀	0.675	0.470	1.000	0.000	1.000
BMI ₀	35.831	5.087	34.816	28.441	60.221
$together_{A}^{+}$	0.656	0.477	1.000	0.000	1.000
BMI_{4}^{+}	34.153	5.149	33.391	22.857	60.672
success ⁺	0.624	0.486	1.000	0.000	1.000
instrumental variables:					
incentive 150 ₁₋₄	0.204	0.404	0.000	0.000	1.000
incentive 3001-4	0.236	0.426	0.000	0.000	1.000
incentive 2505-10	0.312	0.465	0.000	0.000	1.000
incentive 5005-10	0.357	0.481	0.000	0.000	1.000
pharmacy nearby $\frac{1}{1}$	0.682	0.467	1.000	0.000	1.000
			Malac		
dopondont variables:			wates		
dependent variables.	0.806	0.307	1.000	0.000	1 000
sexpuriner11-22	1 515	0.397	2,000	0.000	2 000
sexfrequency ₁₁₋₂₂	0.682	0.091	2.000	0.000	2.000
explanatory variables:	0.065	0.400	1.000	0.000	1.000
ΔBMI_{1-10}	-1.256	2.676	-1.162	-15.347	3.774
$\Delta BMI_{1,22}^+$	-0.526	3.362	-0.299	-18.114	5.495
$\Delta BMI_{\pm 10}^{1-22}$	0.498	1.573	0.393	-8.424	4.181
age	50.183	7.601	50.000	21.000	68.000
female	0.000	0.000	0.000	0.000	0.000
together ₀	0.731	0.446	1.000	0.000	1.000
BMI0	35.164	4.793	34.414	28.441	50.039
together +	0.692	0.464	1.000	0.000	1.000
BMI_{4}^{+}	33.411	4.634	32.747	22.857	48.001
$success^+$	0.635	0.484	1.000	0.000	1.000
instrumental variables:					
incentive 1501-4	0.173	0.380	0.000	0.000	1.000
incentive 3001-4	0.279	0.451	0.000	0.000	1.000
incentive 2505-10	0.308	0.464	0.000	0.000	1.000
incentive 5005-10	0.327	0.471	0.000	0.000	1.000
pharmacy nearby ⁺ ₁₋₂₂	0.663	0.475	1.000	0.000	1.000
			Famalas		
dependent variables:			Tentales		
sernartnera an	0.528	0 504	1.000	0.000	1 000
carfraguancius	1.028	0.004	1.000	0.000	2.000
together ⁺	0.547	0.515	1.000	0.000	2.000
explanatory variables	0.047	0.505	1.000	0.000	1.000
A BML 10	-1 177	2 455	-0.830	_7 249	3 048
ΔBMI^+	0.090	3.182	0.033	-6.732	10.040
ΔBMI_{1-22}	0.351	1.676	0.137	-3 366	4 147
2 Divin ₅₋₁₀	50 132	8 757	51,000	21 000	63 000
uge famala	1 000	0.000	1,000	1 000	1 000
together	0.566	0.000	1.000	0.000	1.000
DMI	27 140	5.300	25 417	20.714	60 221
bivin ₀	0 595	0.407	1 000	29.714	1 000
$logeiner_4$	0.363	0.497	1.000	0.000	1.000
	35.011	5.8U8 0.404	34.297	27.690	00.672
instrumental variables:	0.604	0.494	1.000	0.000	1.000
incentive 150.	0 264	0.445	0.000	0.000	1 000
incentive 1001-4	0.204	0.110	0.000	0.000	1 000
incentive 3001-4	0.131	0.301	0.000	0.000	1.000
incentive 2005-10	0.321	0.4/1	0.000	0.000	1.000
mcentice 5005-10	0.413	0.497	0.000	0.000	1.000
pharmacy nearby 1-22	0.717	0.455	1.000	0.000	1.000

Table 2: Descriptive Statistics for Estimation Sample by Gend	e
---	---

Notes: Statistics for 157 individuals who provide information on *sexpartner*₁₁₋₂₂ or *sexfrequency*₁₁₋₂₂. + Variable used in alternative model specification/robustness check.

left-hand-side variable, generating a spurious correlation between the regressand and ΔBMI_{1-10} . Given the lack of information about how the respondents' sex life develops over time, instrumental variables estimation is the first choice for identifying a causal relation. As the data used in this analysis originates from a randomized field experiment described in section 2.1, it includes variables well suited for being used as instrumental variables. More precisely, these variables are incentive 150₁₋₄, incentive 300₁₋₄, incentive 250₅₋₁₀, and incentive 500₅₋₁₀, i.e. the indicators for randomly assigned group membership in the two intervention phases of the experiment. While exogeneity is guaranteed by the experimental design¹³, explanatory power for the endogenous regressor ΔBMI_{1-10} has been established elsewhere in the literature (Augurzky et al., 2012, 2014). In other words, the analyses cited above clearly find that being exposed to financial incentives for weight loss makes obese individuals loosing body weight. Table 2 displays descriptive statistics for the group indicators (*incentive* 150_{1-4} , ..., *incentive* 500_{5-10}) for males and females, and for the pooled sample. Though the randomization design allocated equal probability to each experimental group, the mean values deviate substantially from 1/3. This can be explained by two reasons: (i) Drop-out from the experiment was more frequent in the control groups where no financial rewards can be earned; (ii) individuals who did not (sufficiently) comply with contractual the weight loss after four months were – without randomization – directly assigned to the no incentive group in the weight maintenance phase.

3 Estimation Procedure

The econometric model rests on a linear equation that links a latent dependent variable to the key regressor ΔBMI_{1-10} , the control variables, and a normally distributed error.¹⁴ This leads to a conventional binary probit model for explaining *sexpartner*₁₁₋₂₂ and an ordered probit model for explaining *sexfrequency*₁₁₋₂₂.

As discussed above, the potential endogeneity of ΔBMI_{1-10} is the key challenge to the empirical analysis that is addressed via the use of instrumental variables (IV). Since the latent dependent variables are not observed, conventional linear instrumental variables estimation, such as two-step least squares, is not an obvious choice for the estimation procedure. For this reason we rely on the assumption of joint normality for identification and specify a control function model.

¹³See section 5 for further discussion of instrument validity.

¹⁴These latent counterparts to *sexpartner*₁₁₋₂₂ and *sexfrequency*₁₁₋₂₂ could be labelled 'inclination and opportunity to live in sexual relationship' and 'inclination and opportunity to have sexual intercourse', respectively. It is important to note that, unlike other behaviors, it is not just individual desire for sex that makes an individual sexually active but also his or her attractiveness to potential partners (and the availability of the latter). Since we cannot distinguish between both in the empirical analysis, we abstain from formulating a more complex econometric model that explicitly takes account of these different aspects.

The control function is equivalent to the instrumental equation in the linear IV-model besides that joint normality is explicitly assumed and used for identification. This distributional assumption allows for joint estimation of all model parameters using maximum likelihood (ML). That means, estimation is carried out in a single step. Deriving a log-likelihood function that is easily dealt with in the optimization procedure rests on expressing the joint likelihood as the product of (i) the probability of the observed outcome conditional on the endogenous regressor and (ii) the density of the endogenous regressor conditional on the instruments. See Wooldridge (2002, p. 476) for ML estimation of the binary probit model with an endogenous regressor. This control function approach is straight forward generalized to related models such as ordered probit required for estimation, individuals for which the dependent variables are not observed but the endogenous regressor is observed contribute to the log-likelihood function and enter the estimation sample. Roodman (2011) discusses various control function models that involve a linear index and jointly normal errors and introduces a software component that implements these models.¹⁵

One may think of estimating the model more conventionally using two-stage least squares, in order to avoid strong distributional assumptions. For the model explaining *sexpartner*₁₁₋₂₂ this is instrumental variables estimation of a linear probability model.¹⁶ For the model explaining *sexfrequency*₁₁₋₂₂ this approach is less straight forward as it requires interpreting the ordered categorial left-hand-side variable to be cardinal. For this reason we stick to the parametric non-linear model as our preferred specification. Yet, in the Appendix we also report results for the linear model, which in qualitative terms do not differ much from their probit counterparts; see Tables 15, 16, 17, and 18.

4 Estimation Results for the Basic Model

In this section we discuss estimation results for the models described above. Table 3 displays results for the model variant that uses *sexpartner*₁₁₋₂₂ as dependent variable, while Table 4 displays results for the model that explains *sexfrequency*₁₁₋₂₂. Besides a pooled model that considers men and women, we discuss results for a stratified specification, as the determinants of sex life are likely to differ substantially between both genders. Starting with the question of what renders a sexual relationship more likely for males, Table 3 (upper panel, left columns) indicates that age does not have a significant effect. Not surprisingly, those who lived with a partner at time zero

 $^{^{15}}$ This Stata[®] ado-file, called *cmp* (conditional mixed process), is used in the present application.

¹⁶However, the linear probability model is – implicitly – subject to strong distributional assumptions too, and has therefore been criticized in the literature (e.g. Horrace and Oaxaca, 2006).

	Mal	les	Females		All		
	Est. Coef.	S.E.	Est. Coef.	S.E.	Est. Coef.	S.E.	
main equation (dependent variable: <i>sexpartner</i> ₁₁₋₂₂)							
ΔBMI_{1-10}	-0.194^{**}	0.086	-0.007	0.212	-0.166^{*}	0.096	
age	0.009	0.020	-0.009	0.023	-0.007	0.015	
female	-	-	-	-	-0.672^{**}	0.254	
together ₀	1.313**	0.381	1.511**	0.555	1.508**	0.280	
BMI ₀	-0.053^{*}	0.030	0.002	0.036	-0.045^{*}	0.025	
constant	1.212	1.533	-0.365	1.745	1.675	1.194	
control function (dependent variable: ΔBMI_{1-10})							
incentive 150 ₁₋₄	-0.701^{*}	0.410	-0.806^{+}	0.511	-0.762^{**}	0.328	
incentive 300 ₁₋₄	-0.960**	0.408	-1.231^{**}	0.484	-1.077^{**}	0.322	
incentive 250 ₅₋₁₀	-2.753^{**}	0.412	-1.318^{**}	0.499	-2.289^{**}	0.322	
incentive 500 ₅₋₁₀	-2.049^{**}	0.402	-1.784^{**}	0.527	-1.877^{**}	0.326	
age	0.004	0.021	-0.007	0.023	0.001	0.016	
female	-	-	-	-	0.076	0.271	
together ₀	-0.485	0.381	0.266	0.420	-0.167	0.290	
BMI ₀	-0.082^{**}	0.028	0.012	0.029	-0.046^{**}	0.021	
constant	3.510**	1.640	-0.152	1.643	2.052*	1.194	
ancillary parameters							
σ (control function error S.D.)	2.654**	0.114	2.256**	0.134	2.556**	0.090	
ρ (error correlation)	0.536**	0.248	-0.443	0.434	0.268	0.276	
# of observations (over all)	27	1	14	1	412	2	
# of observations (main equation)	10	3	53	3	156	6	
log likelihood	-68	4.0	-33	9.5	-103	4.1	
instrument relevance (<i>p</i> -value, joint sig. instruments)	0.00	00	0.00	00	0.00	00	
sample split (p-value, LR-test)	-		-	_		0.069	

Table 3: Involved in a Sexual Relationship ((estimated	coefficients)
--	------------	---------------

Notes: ** significant at 5%; * significant at 10%; + significant at 15%.

(experiment start) are more likely to have a sexual partner one or two years later. Estimation results yield also a significantly negative association with initial BMI. Yet, this coefficient does not isolate a causal relationship but is likely to also capture the influence of unobserved individual heterogeneity.

The key coefficient is rather attached to ΔBMI_{1-10} and is highly significant (*p*-value: 0.025) and negative. This indicates that – conditional on initial weight and initial relationship status – loosing body weight increases the likelihood of being involved in a sexual relationship. The point estimate corresponds to an effect of substantial size. At its maximum value ($\hat{\beta}_{\Delta BMI_{1-10}} \cdot \phi(0)$) the corresponding marginal effect is as high as –0.077. That is one BMI unit reduction in body weight increases the probability of having a sexual partner by 7.7 percentage points. The mean marginal effect in the estimation sample is still 4.1 percentage points.

The estimated error correlation is 0.536 and is statistically significant. From a purely technical perspective, this does not take one by surprise. Since the bivariate descriptive analysis yields a very weak correlation of *BMI*₁₋₁₀ and *sexpartner*₁₁₋₂₂ while the control function approach yields a strong negative effect, something needs to balance the latter for consistency of both results. From a non-technical perspective, it conflicts with our earlier reasoning that self-confidence is a major source of unobserved heterogeneity. It rather argues in favor of the desire for physical pleasures driving the correlation. In other words, those who have a great desire for such pleasures are more

	Mal	es	Females		All	
	Est. Coef.	S.E.	Est. Coef.	S.E.	Est. Coef.	S.E.
main equation (dependent variable: <i>sexfrequency</i> ₁₁₋₂₂)						
ΔBMI_{1-10}	-0.033	0.090	0.059	0.198	-0.057	0.083
age	-0.036^{**}	0.018	-0.011	0.022	-0.025^{*}	0.013
female	-	_	-	_	-0.503^{**}	0.212
together ₀	0.677**	0.276	1.458**	0.501	1.065**	0.221
BMI ₀	-0.050^{*}	0.026	-0.018	0.034	-0.045^{**}	0.021
control function (dependent variable: ΔBMI_{1-10})						
incentive 150 ₁₋₄	-0.817^{**}	0.408	-0.930^{*}	0.544	-0.781^{**}	0.326
incentive 300 ₁₋₄	-0.987^{**}	0.412	-1.256^{**}	0.483	-1.083^{**}	0.322
incentive 250 ₅₋₁₀	-2.680^{**}	0.421	-1.264^{**}	0.518	-2.262**	0.322
incentive 500 ₅₋₁₀	-2.077^{**}	0.405	-1.777^{**}	0.528	-1.903^{**}	0.326
age	0.004	0.021	-0.006	0.023	0.001	0.016
female	-	-	-	-	0.075	0.271
together ₀	-0.491	0.381	0.256	0.421	-0.169	0.290
BMI ₀	-0.083^{**}	0.028	0.012	0.029	-0.047^{**}	0.021
constant	3.607**	1.639	-0.141	1.644	2.070*	1.194
ancillary parameters						
threshold 1	-4.364^{**}	1.420	-0.751	1.623	-3.212**	1.022
threshold 2	-3.362**	1.388	-0.195	1.607	-2.394^{**}	1.012
σ (control function error S.D.)	2.654**	0.114	2.257**	0.135	2.556**	0.089
ρ (error correlation)	0.266	0.254	-0.431	0.434	0.173	0.229
# of observations (over all)	27	1	14	1	412	2
# of observations (main equation)	99)	53	3	152	2
log likelihood	-72	7.5	-35	7.1	-109	8.7
instrument relevance (<i>p</i> -value, joint sig. instruments)	0.00	00	0.00	00	0.00	00
sample split (p-value, LR-test)	-		-		0.005	

Table 4: Frequency	of Sexual Intercourse (estimated coefficients)

Notes: ** significant at 5%; * significant at 10%; + significant at 15%.

likely to have sex but are less likely to reduce overeating. In any case, the estimate for ρ indicates that the endogeneity of ΔBMI_{1-10} is an issue worth to be addressed in the estimation procedure. Interpreting the estimated coefficient in terms of a causal effect critically relies on appropriate instruments that only enter the control function and have substantial explanatory power. Table 3 indicates that the indicators for the experimental groups are indeed strong predictors for the endogenous regressor weight change. The *p*-value for the test on joint significant is smaller than $5 \cdot 10^{-4}$ and each indicator is individually significant in statistical terms. That is, financial incentives do have a strong effect on success in weight loss, as established elsewhere in the literature (Volpp et al., 2008; John et al., 2011; Augurzky et al., 2012; Cawley and Price, 2013; Augurzky et al., 2014), and there is no reason for being concerned about weak instruments.¹⁷

In order to dig deeper into the interdependence of body weight an sexual activity we reestimated the model with an indicator for living together with a partner (*together*₀) as alternative dependent variable, see Appendix Tables 11 and 12. There, we find virtually no effect of a change in body weight. In other words it is not the relationship status per se but its sexual nature that is significantly affected by loosing body weight. In the light of this result one would expect to find also an effect of a change in body weight on the frequency of sexual intercourse. Yet, Table 4 indicates that this does not apply. The point estimate is virtually zero and is accompanied by a

¹⁷In terms of the corresponding linear model the relevant *F*-statistic is 17.812 for males.

large standard error. This raises some concerns with respect to our earlier result of a significant effect on *sexpartner*₁₁₋₂₂.

Turning to the results for females, the estimated effect of ΔBMI_{1-10} on *sexpartner*₁₁₋₂₂ is statistically insignificant. Though the point estimate bears the same sign as its counterparts for males, it is very small and is accompanied by a rather large standard error. One cannot rule out that the effect of body weight on sexual activity is of qualitatively different nature for females as compared to males. Yet, a more trivial explanation for the heterogeneity in results across genders is the small size of the female sample. It includes only 53 individuals for which information on sexual activity is available and, hence, provides a rather weak basis for establishing a causal relationship. Hence it does not take one by surprise that for females significant effects of ΔBMI_{1-10} are also neither found on *sexfrequency*₁₁₋₂₂ nor on *together*₂₂. For females, the estimate for ρ exhibits the opposite sign but is statistically insignificant, raising further doubts with respect to the reliability of the results found for women.

In order to base the econometric analysis on a larger sample, one may think of pooling males and females. Re-estimating the model using the pooled model yields results which are very close and qualitatively equivalent to what is found for males, which dominate the pooled sample in numbers. However, likelihood ratio tests (*p*-values: 0.069, 0.005) argue against the hypothesis that the dependent variables are determined by models that are uniform across gender.

5 Model Extensions and Robustness Checks

Having discussed the results for the basic model, we now turn to two additional specifications that address two potential shortcomings of the models discussed above. In Subsection 5.1 we discuss potential invalidity of some of the instruments used and propose alternative specifications, which use fewer instruments for which validity is no reason for concern. In Subsection 5.2 - in addition to what is discussed in Subsection 5.1 - we discuss potential selection bias and propose a model variant that corrects for it.

5.1 Analysis Conditional on Success in Weight-Loss Phase

Though the parametric control function model is non-linear, non-linearity is not sufficient for identification which still rests on valid exclusion restrictions. In the above section, we argued that group membership is purely random and exerts effects on sex life only through the change in body weight. While – by experimental design – randomness applies to group membership in

the weight-loss phase, one may however put randomness into question with respect to the group membership in the weight-maintenance phase that is for the indicators *incentive* 250_{5-10} and *incentive* 500_{5-10} . This is due to the experimental design that conditions assignment to the intervention groups in the weight-maintenance on success in the preceding phase. In other words, one may argue that the effect of unobserved confounding factors is not be completely eliminated by using *incentive* 150_{1-4} , *incentive* 300_{1-4} , *incentive* 250_{5-10} , and *incentive* 500_{5-10} as instruments, since the latter two are potentially affected by these factors through being successful in loosing body weight during the months one to four.

One obvious approach to deal with this issue is to use only *incentive* 150_{1-4} and *incentive* 300_{1-4} as instruments for ΔBMI_{1-10} . However, as compared to *incentive* 250_{5-10} and *incentive* 500_{5-10} , group membership in the weight-loss phase has less explanatory power for the endogenous regressor. That is, group membership in the weight-reduction phase is just a weak instrument for weight loss over the entire intervention period of the experiment. This can easily be explained by long-term effects of financial incentives – if existent – being much smaller than short-term effects (Augurzky et al., 2014). In addition, and even more important, a significant reduced form effect of *incentive* 150_{1-4} and *incentive* 300_{1-4} is not found in the data, neither on *sexpart*₁₁₋₂₂ nor on *sexfrequency*₁₁₋₂₂. In other words identification – even in the basic model – critically rests on *incentive* 250_{5-10} , and *incentive* 500_{5-10} being used as instruments.

For this reason we take a different approach to tackling the potential endogeneity of the instruments by conditioning the analysis on success in the weight-reduction phase, i.e. by conditioning on eligibility for the second randomization. We implement this approach in two different ways. The first is (i) confining the analysis to those individuals who did successfully loose weight in the weight-loss phase and, hence, are eligible for the second randomization. Within this sub-sample group membership from month 5 to month 10 is purely random and, in turn, incentive 2505-10 and *incentive* 500₅₋₁₀ are purely exogenous with respect to weight change over the weight maintenance phase (ΔBMI_{5-10}). This does not hold with respect to weight change over the entire intervention period (ΔBMI_{1-10}). For this reason, in this alternative model that only considers individuals who were successful in the first intervention period ΔBMI_{5-10} serves as key regressor and *incentive* 250_{5-10} and *incentive* 500_{5-10} as instruments for it. In this model *incentive* 150_{1-4} and *incentive* 300_{1-4} are no valid instruments. One may think of including them as further control variables to the model. Yet - due to randomization - in the considered sub-sample the weightmaintenance incentives are orthogonal to the weight-loss incentives and controlling for possible long-run effects of exposition to the latter is not essential for identification. Hence, for the sake of a parsimoniously specified model, we exclude incentive 1501-4 and incentive 3001-4 from the regres-

	Males		Females		All	
	Est. Coef.	S.E.	Est. Coef.	S.E.	Est. Coef.	S.E.
main equation (dependent variable: <i>sexpartner</i> ₁₁₋₂₂)						
ΔBMI_{5-10}	-0.403^{**}	0.087	0.422**	0.118	-0.368^{**}	0.154
age	-0.003	0.016	0.023^{+}	0.015	-0.015	0.014
female	-	-	-	-	-0.325	0.310
together ₄	0.738^{*}	0.384	-0.005	0.660	1.133**	0.474
BMI4	-0.034	0.029	-0.045^{*}	0.026	-0.045	0.033
constant	1.549	1.383	0.389	1.120	2.369*	1.372
control function (dependent variable: ΔBMI_{5-10})						
incentive 250 ₅₋₁₀	-1.409^{**}	0.396	-0.191	0.447	-1.040^{**}	0.325
incentive 500 ₅₋₁₀	-1.015^{**}	0.369	-0.423	0.611	-0.719^{**}	0.326
age	0.004	0.020	-0.049^{*}	0.027	-0.014	0.016
female	-	_	-	_	0.117	0.291
together ₄	-0.103	0.378	0.799^{+}	0.518	0.266	0.310
BMI4	-0.031	0.029	0.079**	0.036	0.012	0.023
constant	2.095	1.501	-0.288	1.811	1.003	1.177
ancillary parameters						
σ (control function error S.D.)	2.116**	0.112	2.085**	0.164	2.150**	0.094
ρ (error correlation)	0.851**	0.142	-0.976^{**}	0.080	0.687**	0.347
# of observations (over all)	179		81		260)
# of observations (main equation)	65		32	2	97	
log likelihood	-41	0.1	-18	8.0	-605	5.7
instrument relevance (<i>p</i> -value, joint sig. instruments)	0.00)1	0.77	73	0.00	15
sample split (<i>p</i> -value, <i>LR</i> -test)	-		-		0.17	0

Table 5: Sexual Relationship, cond. on success in weight-loss phase (est. coefficients)

Notes: ** significant at 5%; * significant at 10%; ⁺ significant at 15%. All model equations estimated using only individuals who successfully lost body weight during month one to four.

sion model.¹⁸ Moreover, as in this variant of the model identification rests exclusively on weight variation over the maintenance phase, we condition on the values of the control variables (initial *BMI*, relationship status) four months after experiment start.

Confining the analysis to individuals who are eligible to the second randomization reduces the size of the estimation sample considerably.¹⁹ The second model variant addresses this shortcoming by (ii) estimating the model described above using the entire sample, i.e. including those individuals who did not qualify for the second randomization. Yet, estimation is still conditional on the weight-outcome of the weight-loss phase since a further binary control variable *success*₄ is included, indicating eligibility for the second randomization four month after rehab discharge. The reasoning behind this specification is that conditional on failure or success the instruments *incentive* 250₅₋₁₀ and *incentive* 500₅₋₁₀ are exogenous. However *success*₄ is certainly a 'bad control' as it itself represents an outcome variable closely related to the endogenous regressor ΔBMI_{5-10} . We address this by augmenting the model by a further control function that explains *success*₄ and uses *incentive* 150₁₋₄ and *incentive* 300₁₋₄ as instruments. As discussed in Augurzky et al. (2012), the weight-loss incentives have strong explanatory power for success in the weight-loss phase of the experiment.

¹⁸Letting *incentive* 150₁₋₄ and *incentive* 300₁₋₄ enter the regression model as further controls has little effect on the estimate for the key coefficient.

¹⁹The value for N is reduced to 260 as compared to 412 for the basic model (pooled sample); see Tables 3 and 5.

	Males		Females		All	
	Est. Coef.	S.E.	Est. Coef.	S.E.	Est. Coef.	S.E.
main equation (dependent variable: <i>sexfrequency</i> ₁₁₋₂₂)						
ΔBMI_{5-10}	-0.275^{*}	0.146	0.462**	0.091	-0.268^{+}	0.186
age	-0.023	0.019	0.022	0.016	-0.026^{*}	0.014
female	-	-	-	-	-0.115	0.249
together ₄	0.519*	0.313	-0.133	0.967	0.945**	0.348
BMI_4	-0.045^{+}	0.030	-0.047	0.039	-0.049^{+}	0.031
control function (dependent variable: ΔBMI_{5-10})						
incentive 2505-10	-1.280^{**}	0.440	0.028	0.189	-0.980^{**}	0.346
incentive 500 ₅₋₁₀	-1.170^{**}	0.358	-0.171	0.758	-0.833^{**}	0.310
age	0.003	0.020	-0.050^{*}	0.027	-0.014	0.016
female	-	-	-	-	0.108	0.291
together ₄	-0.106	0.379	0.765^{+}	0.516	0.262	0.311
BMI4	-0.033	0.029	0.078**	0.036	0.011	0.023
constant	2.236^{+}	1.502	-0.298	1.818	1.056	1.176
ancillary parameters						
threshold 1	-3.245^{*}	1.730	-0.498	1.651	-3.200**	1.436
threshold 2	-2.528^{+}	1.546	-0.435	1.444	-2.558^{**}	1.273
σ (control function error S.D.)	2.120**	0.112	2.090**	0.164	2.151**	0.094
ho (error correlation)	0.750**	0.233	-0.993**	0.061	0.627*	0.368
# of observations (over all)	17	9	81		260	0
# of observations (main equation)	63	3	32		95	
log likelihood	-43	7.9	-19	6.6	-64	5.6
instrument relevance (<i>p</i> -value, joint sig. instruments)	0.00	02	0.93	39	0.00)6
sample split (<i>p</i> -value, <i>LR</i> -test)	_		_		0.015	

TT 1 1 /	T.	(T)	1 •	• 1 • 1	1 (, (,)
I abio 6	· Hroanov	7 At Intorcource	cond on success 1	n_{11}	nhacoloct cootticiontel
I adde u	. FICUUCIUV	OI IIIICICOUISC	, conu, on success i		

Notes: ** significant at 5%; * significant at 10%; + significant at 15%. All model equations estimated using only individuals who successfully lost body weight during month one to four.

Tables 5 and 6 display estimation results for the variant of the model that confines the analysis to those, who successfully lost weight over month one to month four. As before, besides estimating a pooled model, we estimate the model stratified by gender. Here, the females' regression clearly does not yield reliable results for the causal effect of body weight on women's sex life. The reason for this is the very low explanatory power the instruments have for ΔBMI_{5-10} . The *p*-value for the test on joint significant is as high as 0.773 indicating that the incentive indicators are weak instruments in the females' sub-sample. Thus, the puzzling positive and – in the regression explaining *sexpartner*₁₁₋₂₂ – significant coefficient of ΔBMI_{5-10} represents an artifact of weak instruments.

This argument does not apply to the male's sub-sample and the pooled sample, see Tables 5 and 6. There, the tests on the joint significance of *incentive* 250_{5-10} and *incentive* 500_{5-10} yield much smaller *p*-values (0.001 and 0.002). The coefficients attached to ΔBMI_{5-10} – in qualitative terms – confirm what is found in the specification that uses weight change over the entire intervention period as explanatory variable. That is weight-loss significantly increases the probability for being involved in a sexual relationship for males and in the pooled sample. Yet, the coefficients are much larger in magnitude as compared to Table 3 and 4. Marginal effects on the probability of *sexpartner*₁₁₋₂₂ taking the value of one are more than two times larger, reaching the value of -0.161 at its maximum. In other words, loosing one BMI unit in bodyweight makes an sexual

	Males		Females		All	
	Est. Coef.	S.E.	Est. Coef.	S.E.	Est. Coef.	S.E.
main equation (dependent variable: <i>sexpartner</i> ₁₁₋₂₂)						
ΔBMI_{5-10}	-0.425^{**}	0.166	-0.004	0.107	-0.386**	0.150
age	-0.007	0.016	-0.011	0.018	-0.020^{*}	0.012
female	-	_	-	_	-0.690**	0.289
together ₄	0.579**	0.289	1.561**	0.483	1.233**	0.484
BMI4	-0.039	0.029	-0.000	0.032	-0.027	0.023
SUCCESS4	0.549	0.951	-0.564^{*}	0.316	-0.612	0.627
constant	1.732	1.240	-0.113	1.423	2.391**	0.957
control function (dependent variable: ΔBMI_{5-10})						
incentive 250 ₅₋₁₀	-1.382^{**}	0.396	-0.681	0.510	-1.018^{**}	0.295
incentive 500 ₅₋₁₀	-0.962^{**}	0.372	-0.689	0.541	-0.731^{**}	0.300
age	-0.012	0.020	-0.045^{**}	0.021	-0.028^{**}	0.013
female	-	-	-	-	-0.288	0.222
together ₄	-0.077	0.311	0.779**	0.375	0.276	0.237
BMI ₄	-0.002	0.023	0.032	0.027	0.009	0.017
success ₄	-0.924	1.022	-0.443	0.453	-1.086	0.816
constant	2.281^{+}	1.444	1.293	1.519	2.384**	1.039
control function (dependent variable: <i>success</i> ₄)						
incentive 150 ₁₋₄	0.599**	0.165	0.369^{+}	0.252	0.559**	0.145
incentive 300 ₁₋₄	0.562**	0.166	0.833**	0.234	0.650**	0.146
age	0.000	0.009	-0.015	0.012	-0.006	0.007
female	-	-	-	-	-0.179^{+}	0.122
together ₀	0.373**	0.165	0.253	0.214	0.260**	0.130
BMI ₀	0.013	0.013	-0.005	0.016	0.006	0.010
constant	-0.840	0.725	0.467	0.834	-0.245	0.532
ancillary parameters						
σ (control function error S.D.)	1.986**	0.141	2.066**	0.124	2.096**	0.159
ρ_{bmi}^{sex}	0.641^{*}	0.379	-0.243	-	0.707**	0.311
$\rho_{success}^{sex}$	-0.413	0.540	0.772**	0.037	0.498	0.365
$ \rho_{success}^{bmi} $	0.435*	0.258	0.429	-	0.486**	0.205
# of observations (over all)	342	7	17	2	519	9
# of observations (main equation)	103	3	53	3	150	6
log likelihood	-80	8.3	-42	0.8	-124	6.2
instrument relevance (p-value, joint sig. instruments)	0.00)2	0.30)7	0.00)2
sample split (p-value, LR-test)	-		-		0.01	17

\mathbf{T}	Table 7: Sexual Relationsh	ip, control for success	in weight-loss p	hase (est. coef.)
--------------	----------------------------	-------------------------	------------------	-------------------

Notes: ** significant at 5%; * significant at 10%; $^+$ significant at 15%.

relationship substantially more likely. Estimating the model on basis of the pooled sample yields a similar result.

Unlike the results discussed in section 4, a significant – at the ten percent level – and negative effect of ΔBMI_{5-10} is also found for *sexfrequency*₁₁₋₂₂, see Table 6. That is, estimating the model conditional on success in the weight-loss phase yields not only the result that loosing body weight makes a sexual relationship more likely but also that having actually sex becomes more frequent if obese individuals become less overweight. One may explain this finding, which deviates from the results discussed in Section 4, by relying on more credible exclusion restrictions. An alternative explanation could be that adjusting sex life habits (in an existing partnership) takes some time and reduced body weight comes only in effect if it is maintained for some time. Due to the ordinal nature of *sexfrequency*₁₁₋₂₂ the coefficient is not easily interpreted in qualitative terms. Yet, one may interpret the coefficient such that if the odds for having sex at least as frequent as indicated by the *k*th category of *sexfrequency*₁₁₋₂₂ are 50 percent, a one BMI unit reduction in body weight

	Mal	es	Females		All	
	Est. Coef.	S.E.	Est. Coef.	S.E.	Est. Coef.	S.E.
main equation (dependent variable: <i>sexfrequency</i> ₁₁₋₂₂)						
ΔBMI_{5-10}	-0.296^{*}	0.175	0.392**	0.077	-0.268^{+}	0.183
age	-0.031^{**}	0.015	0.020	0.018	-0.031^{**}	0.012
female	-	-	-	-	-0.533^{**}	0.211
together ₄	0.416^{+}	0.262	0.204	1.156	0.979**	0.308
BMI4	-0.040^{+}	0.025	-0.016	0.021	-0.031^{+}	0.021
success ₄	0.084	0.794	0.938^{+}	0.615	-0.829^{+}	0.531
control function (dependent variable: ΔBMI_{5-10})						
incentive 250 ₅₋₁₀	-1.322^{**}	0.376	-0.227	0.441	-0.973^{**}	0.308
incentive 5005-10	-1.075^{**}	0.328	-0.423	0.546	-0.822^{**}	0.288
age	-0.013	0.016	-0.059^{**}	0.024	-0.028^{**}	0.013
female	-	-	-	-	-0.294	0.222
together ₄	-0.075	0.286	0.891*	0.461	0.277	0.236
BMI4	-0.004	0.022	0.025	0.031	0.009	0.017
success ₄	-0.772	0.883	-2.566^{**}	0.916	-1.097	0.788
constant	2.294*	1.233	3.250*	1.721	2.408**	1.034
control function (dependent variable: <i>success</i> ₄)						
incentive 150 ₁₋₄	0.661**	0.167	0.390^{+}	0.246	0.553**	0.143
incentive 3001-4	0.555**	0.166	0.571**	0.243	0.647**	0.144
age	0.000	0.009	-0.020^{+}	0.013	-0.006	0.007
female	-	-	-	-	-0.177^{+}	0.122
together ₀	0.324**	0.162	0.161	0.218	0.264**	0.129
BMI ₀	0.013	0.013	0.001	0.016	0.007	0.010
constant	-0.823	0.703	0.614	0.824	-0.255	0.531
ancillary parameters						
threshold 1	-3.717^{**}	1.378	1.035	1.236	-3.475^{**}	0.976
threshold 2	-2.956^{**}	1.271	1.241	0.979	-2.829^{**}	0.880
σ (control function error S.D.)	1.977**	0.146	2.521**	0.240	2.101**	0.156
ρ_{bmi}^{sex}	0.558^{+}	0.356	-0.942^{**}	0.241	0.607*	0.357
$\rho_{success}^{sex}$	-0.286	0.479	-0.613	0.553	0.519^{+}	0.334
$ \rho_{success}^{bmi} $	0.384^{+}	0.246	0.826**	0.113	0.493**	0.196
# of observations (over all)	342	7	172	2	519	9
# of observations (main equation)	99		53		152	2
log likelihood	-85	1.6	-43	6.3	-131	0.6
instrument relevance (<i>p</i> -value, joint sig. instruments)	0.00)1	0.73	34	0.00)2
sample split (<i>p</i> -value, <i>LR</i> -test)	-		-		0.00	00

Notes: ** significant at 5%; * significant at 10%; + significant at 15%.

increases this probability by 11 percentage points. As before, the results for the pooled sample largely mirror what is found for males.

Turning to the results for the model variant that uses the full sample for estimation but includes *success*₄ as control and adds a second control function, see Tables 7 and 8, they largely mirror what is found for the conditional model discussed above. Results for females still seem not to be reliable because of relatively weak instruments. The key point estimate, as far as the model explaining *sexpartner*₁₁₋₂₂ is concerned, indeed differs much from the corresponding estimate result reported in Table 5, shedding further doubts on the results found for women. This does not apply to the males' and pooled sample. There the estimated coefficients of ΔBMI_{5-10} are very similar to their counterparts from the model that only considers successful individuals. This also applies to the estimated standard errors and, in turn, to the levels of statistical significance. This corresponds to both variants of the model relying on variation in ΔBMI_{5-10} within the group of successful weight losers for identifying the effect of weight change on sexual activity. It nevertheless provides more credence in the robustness of the results to changes in the model specification.

5.2 Selection Correction

One major concern about the present analysis is, as compared to the initial study population, the small size of the actual estimation sample. In other words, the experiment suffers from severe sample attrition, which may not be exogenous and, in turn, may be a source of bias. Besides randomly removing all questions concerning sexual behavior from roughly one-half of the questionnaires – which does not challenge consistency – two distinct processes are responsible for the small number of individuals for whom information about sex life is available. As one can easily imagine, questions concerning the respondents' sexual behavior are delicate and, for this reason, are subject to item non-response. Yet, in our data the rate of denied information is smaller than 0.1, which is remarkably low compared to survey studies that ask for information on sex life (Fenton et al., 2001). This rate applies to those, who have completed the experiment and provided information on body weight at its final stage. Yet - more importantly - roughly 55 percent of the participants dropped out at some point in time and denied providing further information.²⁰ In most cases dropping out took place without explicit notice to experimenters by not showing-up at the weigh-in and not returning the questionnaire, even after receiving reminder letters and reminder phone calls. Taken all sources of missing information together, information on sex life is available for just 22.5 percent of the original study population.

While item non response is not easily addressed in the econometric analysis, the experiment generated information that can be used to address drop-out. Since the show-up fee for attending the weigh-ins is flat and does not account for the actual costs of providing the requested information, the actual net-cost of experiment continuation varies substantially across individuals. A major determinant of costs differentials is group membership as for – successful – members of the incentive groups, dropping out means foregoing financial rewards. Yet, this information is already exhausted for explaining changes in body weight and cannot be used as an instrument for compliance. However, another determinant is travel cost. Though no precise measure for this cost is available – which is highly subjective und inherently unobservable – the data include an indicator for the place of weigh-in is located within the same zip-code area as the participant's place of residence. Since the pharmacy to be visited for the weigh-ins was assigned by the experimenter, this variable (*pharmacy nearby*₁₋₂₂) represents an exogenous source of variation in travel

²⁰With respect to dropping out, body weight is the key piece of information. As providing information about body weight requires attending a weigh-in it is more costly than filling-in a questionnaire, which is sufficient for providing information concerning other variables.

	Ma	les	Female	es	Al	1
	Est. Coef.	S.E.	Est. Coef.	S.E.	Est. Coef.	S.E.
main equation (dependent variable: <i>sexpartner</i> ₁₁₋₂₂)						
ΔBMI_{5-10}	-0.359^{**}	0.164	_	-	-0.322^{*}	0.178
age	0.011	0.025	-	-	-0.000	0.012
female	-	-	-	-	-0.059	0.225
together ₄	0.260	0.299	-	-	0.482**	0.235
BMI4	-0.062^{*}	0.037	-	-	-0.058^{*}	0.034
constant	1.258	1.802	-	-	1.436	1.201
control function (dependent variable: ΔBMI_{5-10})						
incentive 2505-10	-1.399^{**}	0.500	-	-	-1.090^{**}	0.389
incentive 500 ₅₋₁₀	-1.068^{**}	0.479	-	-	-0.750^{**}	0.365
age	0.007	0.028	-	-	-0.013	0.018
female	-	-	-	-	0.112	0.334
together ₄	-0.111	0.418	-	-	0.289	0.323
BMI4	-0.032	0.030	-	-	0.012	0.021
constant	2.038	2.013	-	-	1.004	1.311
selection equation (dependent variable: <i>sexinfo</i> ₂₂)						
pharmacy nearby ₁₋₂₂	0.880**	0.205	-	_	0.858**	0.166
incentive 2505-10	-0.251	0.249	-	-	-0.010	0.205
incentive 500 ₅₋₁₀	0.141	0.252	-	-	0.162	0.196
age	0.020	0.014	-	-	0.014	0.011
female	-	-	-	-	0.110	0.177
together ₄	-0.384^{+}	0.262	-	-	-0.305^{+}	0.200
BMI4	-0.061^{**}	0.027	-	-	-0.050^{**}	0.019
constant	0.486	1.261	-	-	0.279	0.844
ancillary parameters						
σ (control function error S.D.)	2.133**	0.099	-	-	2.167**	0.072
ρ_{hmi}^{sex}	0.689**	0.347	-	-	0.555	0.414
ρ_{info}^{sex}	0.691*	0.389	-	-	0.809**	0.302
$ \rho_{info}^{bmi} $	-0.047	0.144	-	-	-0.039	0.106
# of observations (over all)	21	2	_		30	7
# of observations (main equation)	65	5	-		97	7
log likelihood	-51	5.8	_		-76	3.5
exclusion restriction (p-value, selection equation)	0.0	00	-		0.00	00
sample split (p-value, LR-test)	-		-		0.39	90

Table 9:	Sexual	Relationship,	selection	correction	(estimated	coefficients)
					·	,

Notes: ** significant at 5%; * significant at 10%; + significant at 15%. All model equations estimated using only individuals who successfully lost body weight during month one to four.

cost and can be used as instrument for providing information on sex life at the final stage of the experiment. In oder words, the underlying intuition for using *pharmacy nearby*₁₋₂₂ as instrument is that being sent to a nearby pharmacy is positively correlated with program continuation because of little time- and transportation costs, but it should in no way be related directly to sexual behavior.

In terms of the econometric model we augment the first variant of the model discussed in section 5.1 (only successful participants included) by a selection equation that explains whether or not information on individuals *i*'s sex life is available in the data $(sexinfo_{22})^{21}$ and that includes *pharmacy nearby*₁₋₂₂ as additional explanatory variable.²² Following our previous modelling strategy, we assume joint normality with the main equation and the control function and estimate all

 $^{^{21}}$ Item-non-response slightly varies between the two dependent variables *sexpartner*₁₁₋₂₂ and *sexfrequency*₁₁₋₂₂. Hence, the dependent variable of the selection equation *sexinfo*₂₂ is differently defined for the two variants of the model.

 $^{^{22}}$ One may also augment the model that includes *success*₄ as control by a third control function explaining *sexinfo*₂₂. This would however require simulated ML estimation, involving a substantial computational burden. Hence, given that the two variants of the model discussed in section 5.1 yield very similar results, we focus on the first variant when addressing the problem of possible selection bias.

	Ma	les	Femal	Females		All	
	Est. Coef.	S.E.	Est. Coef.	S.E.	Est. Coef.	S.E.	
main equation (dependent variable: <i>sexfrequency</i> ₁₁₋₂	2)						
ΔBMI_{5-10}	-0.316^{**}	0.131	-	-	-0.331^{**}	0.122	
age	-0.014	0.025	-	-	-0.015	0.012	
female	-	-	-	-	0.013	0.210	
together ₄	0.356	0.318	-	-	0.563**	0.272	
BMI_4	-0.054	0.039	-	-	-0.054^{+}	0.033	
control function (dependent variable: ΔBMI_{5-10})							
incentive 250 ₅₋₁₀	-1.183^{**}	0.495	-	-	-0.922^{**}	0.379	
incentive 500 ₅₋₁₀	-1.210^{**}	0.473	-	-	-0.878^{**}	0.355	
age	0.003	0.028	-	-	-0.014	0.018	
female	-	-	-	-	0.105	0.335	
together ₄	-0.110	0.432	-	-	0.260	0.321	
BMI4	-0.034	0.029	-	-	0.011	0.020	
constant	2.290	1.948	-	-	1.089	1.279	
selection equation (dependent variable: <i>sexinfo</i> ₂₂)							
pharmacy nearby ₁₋₂₂	0.815**	0.220	-	-	0.843**	0.172	
incentive 250 ₅₋₁₀	-0.085	0.280	-	-	0.126	0.209	
incentive 500 ₅₋₁₀	0.213	0.266	-	-	0.125	0.202	
age	0.022^{+}	0.014	-	-	0.014	0.010	
female	-	-	-	-	0.129	0.181	
together ₄	-0.296	0.278	-	-	-0.229	0.207	
BMI4	-0.063^{**}	0.028	-	-	-0.053^{**}	0.019	
constant	0.312	1.334	-	-	0.247	0.855	
ancillary parameters							
threshold 1	-2.776	1.952	-	-	-2.371^{*}	1.346	
threshold 2	-2.174	1.815	-	-	-1.912^{+}	1.228	
σ (control function error S.D.)	2.124**	0.095	-	-	2.154**	0.069	
ρ_{bmi}^{sex}	0.805**	0.186	-	-	0.747**	0.246	
ρ_{info}^{sex}	0.243	0.371	-	-	0.434^{+}	0.296	
ρ_{info}^{bmi}	-0.061	0.149	-	-	-0.045	0.109	
# of observations (over all)	21	2	_		30	7	
# of observations (main equation)	63	3	_		95	;	
log likelihood	-54	7.2	_		-80	8.6	
exclusion restriction (p-value, selection equation)	0.0	00	_		0.00	00	
sample split (<i>p</i> -value, <i>LR</i> -test)	-		-		0.04	47	

Table 10: Frec	uency of Interco	ourse, selection co	orrection (estimated	l coefficients)
				,

Notes: ** significant at 5%; * significant at 10%; + significant at 15%. All model equations estimated using only individuals who successfully lost body weight during month one to four.

model parameters jointly by maximum likelihood. Estimation results are displayed in the Tables 9 and 10. No results are reported for the females' sub-sample because of weak instruments, already discussed in section 5.1.

The relevant tests indicate that *pharmacy nearby*₁₋₂₂ has substantial explanatory power for *sexinfo*₂₂, i.e. identification rests on a strong exclusion restriction. This is in line with missing information being predominantly due to drop-out from the experiment while item-non-response – which should not be correlated with *pharmacy nearby*₁₋₂₂ – plays just a marginal role. The point estimates for the error correlation ρ_{info}^{sex} are positive, pointing to unobserved factors – such as personal traits – that make individuals sexually more active and, at the same time, more willing to reveal information about their sex lives.

The selection correction model confirms the previous results, and hence provides some confidence in the estimation results not representing an artifact of self-selection. In all variants of the model, the coefficients attached to ΔBMI_{5-10} bear the same sign and are of similar magnitude as their counterparts in Table 5 and Table 6. The largest deviation from the results without selection correction is found for the model explaining *sexpartner*₁₁₋₂₂ only considering males. There, the coefficient gets somewhat smaller in absolute terms while the estimated standard error gets almost twice as large if the selection equation is added to the model. Yet, the relevant *p*-value is still as low as 0.028. The key result that weight loss exerts a significant and positive effect of substantial magnitude on both the probability of sexual relationship and the frequency of intercourse is hence not challenged by correction for endogenous sample selection. This also applies to a slightly different variant of the model that uses the entire sample for estimating the selection equation and not just those individuals who qualify for the second randomization.

6 Discussion and Conclusions

This paper analyzed whether weight reduction in adult obese individuals exerts an effect of their sex lives, namely whether or not they live in a sexual partnership and how frequently they have sex. Based on data generated by a randomized field experiment, where randomization provides an exogenous source of variation in body weight that is exploited for identification, we find that obese males are substantially more likely to be involved in a sexual relationship if they get less overweight. This effect is of substantial magnitude reaching, depending on the model specification, 8 to 17 percentage points for the most effected individual. The results are not as clear for the frequency of sexual intercourse. Yet, in males who have already lost weight, estimation results indicate that further weight reduction – or at least weight maintenance – results in having sex more frequently. For males the key results are robust to several variations to the specification, such as being more restrictive in choosing valid instruments, choosing a different observation period for the key explanatory variable, estimating a linear probability rather than a probit model, and to correcting for experiment-dropout via augmenting the model by a selection equation.

In contrast, for females the analysis does not yield any significant and reliable effect. This can be explained by the small number of women in our data that seriously hampers the empirical analysis. Yet, gender differentials in what determines sex life might provide another explanation. According to Baumeister et al. (2001) there is ample empirical evidence for men – as compared to women – having more frequent and more intense sexual desires. The strength of sexual motivation, also referred to as sex drive, is for instance reflected in the desired number of partners and the desired frequency of intercourse, which are closely related to the outcome variables of the present study. The stronger male sex drive may render detecting a causal effect of weight loss easier for men than for women. Based on an extensive literature survey, Petersen and Hyde (2010) state only small gender differences in many sexual behaviours. Yet, casual sex and attitudes toward casual sex are among the exceptions to this general finding. Regarding these two dimensions men report a more active sexual behavior or express more permissive attitudes towards them than women do. This diverging predisposition could be another explanation for the significant effects found for men but not for women, concerning the probability for being involved in a sexual relationship and the frequency of sexual intercourse

Since for many individuals a fulfilling sex life is an important facet of general life satisfaction and happiness, our key result may provide additional motivation for obese males to reduce overweight, similar to tobacco education where preventing impairments in sexual life is an important argument for smoking cessation (Linnebur, 2006). Moreover, it provides further evidence for substantial social costs of obesity that are not necessarily linked to pathological health problems. This raises the question about the channels through which the established effects operates, which cannot be decided on basis of the data used in this study. One possibility is recovery from sexual dysfunction on which the medical literature has focussed. Yet, given strong effects of a moderate weight reduction found in the data, other channels are likely to also play a role. Greater selfconfidence when initiating sexual contact, improved attractiveness to potential sexual partners, feeling more comfortable with its own body, and more desire for sex may act as such channels. One limitation of the present analysis is that it does not allow for disentangling the effect of a reduced body weight from the effect of the process of weight reduction. The strong effects found may therefore rather be attributed to being successful in meeting a major individual challenge than to weight-loss itself. One possible route to address this question were to evaluate the effects of bariatric surgeries on sexual activity.

While the existing non-medical literature had a strong focus of the effect of overweight on young individuals' sex lives, our result explicitly refers to adult obese individuals. Thus can be interpreted in terms of another desirable consequence of weight reduction. In fact the argument of overweight having the desirable side effect of counteracting premature initiation to sex, put forward elsewhere in the literature, does not apply given the age group considered in the present analysis. Our results rather provides further evidence for obesity imposing restrictions on private lives, but also for even moderate weight reduction mitigating these limitations.

References

- Adolfsson, B., Elofsson, S., Rössner, S. and Undén, A.-L. (2004). Are Sexual Dissatisfaction and Sexual Abuse Associated with Obesity? A Population-Based Study, *Obesity Research* 12(10): 1702–1709.
- Ali, M. M., Rizzo, J. A., Amialchuk, A. and Heiland, F. (2014). Racial differences in the influence of female adolescents' body size on dating and sex, *Economics and Human Biology* **12**: 140–152.
- Augurzky, B., Bauer, T. K., Reichert, A. R., Schmidt, C. M. and Tauchmann, H. (2012). Does Money Burn Fat? Evidence from a Randomized Experiment, *Ruhr Economic Papers* **368**.
- Augurzky, B., Bauer, T. K., Reichert, A. R., Schmidt, C. M. and Tauchmann, H. (2014). Small Cash Rewards for Big Losers - Experimental Insights Into the Fight Against the Obesity Epidemic, *Ruhr Economic Papers* **530**.
- Baumeister, R. F., Catanese, K. R. and Vohs, K. D. (2001). Is there a gender difference in strength of sex drive? Theoretical views, conceptual distinctions, and a review of relevant evidence, *Personality and Social Psychology Review* **5**(3): 242–273.
- Cawley, J. (2001). Body weight and the dating and sexual behaviors of young adolescents, *in*R. T. Michael (ed.), *Social Awakening: Adolescent Behavior as Adulthood Approaches*, Russell Sage,pp. 174–198.
- Cawley, J., Joyner, K. and Sobal, J. (2006). Size Matters: The Influence of Adolescents' Weight and Height on Dating and Sex, *Rationality and Society* **18**: 67–94.
- Cawley, J. and Price, J. A. (2013). A case study of a workplace wellness program that offers financial incentives for weight loss, *Journal of Health Economics* **32**: 794–803.
- Esposito, K., Giugliano, F., Palo, C. D., Giugliano, G., Marfella, R., D'Andrea, F., D'Armiento, M. and Giugliano, D. (2004). Effect of lifestyle changes on erectile dysfunction in obese men: A randomized controlled trial, *JAMA: The Journal of the American Medical Association* 291(24): 2978– 2984.
- Fenton, K., Johnson, A., McManus, S. and Erens, B. (2001). Measuring sexual behaviour: methodological challenges in survey research, *Sexually Transmitted Infections* 77(2): 84–92.
- Halpern, C. T., King, R. B., Oslak, S. G. and Udry, J. R. (2005). Body Mass Index, Dieting, Romance, and Sexual Activity in Adolescent Girls: Relationships Over Time, *Journal of Research on Adolescence* 15(4): 535–559.

- Harris, K. M. (2013). The Add Health Study: Design and Accomplishments, Carolina Population Center, University of North Carolina at Chapel Hill.
 URL: http://www.cpc.unc.edu/projects/addhealth/data/guides
- Horrace, W. C. and Oaxaca, R. L. (2006). Results on the bias and inconsistency of ordinary least squares for the linear probability model, *Economics Letters* **90**: 321–327.
- John, L., Loewenstein, G., Troxel, A., Norton, L., Fassbender, J. and Volpp, K. (2011). Financial Incentives for Extended Weight Loss: A Randomized, Controlled Trial, *Journal of General Internal Medicine* 26: 621–626.
- Kolotkin, R. L., Binks, M., Crosby, R. D., Østbye, T., Gress, R. E. and Adams, T. D. (2006). Obesity and Sexual Quality of Life, *Obesity* **14**(3): 472–479.
- Larsen, S. H., Wagner, G. and Heitmann, B. L. (2007). Sexual Function and Obesity, *International Journal of Obesity* **31**: 1189–1198.
- Linnebur, S. A. (2006). Tobacco education: Emphasizing impotence as a consequence of smoking, *American Journal of Health-System Pharmacy* **63**(24): 2509–2512.
- Neymotin, F. and Downing-Matibag, T. (2014). Adolescent Obesity: A Barrier to Mate Selection?, *Journal of Child & Adolescent Behavior* **2**(4): 1–3.
- Petersen, J. L. and Hyde, J. S. (2010). A meta-analytic review of research on gender differences in sexuality, 1993-2007, *Psychological Bulletin* **136**(1): 21–38.
- Reichert, A. R. (2015). Obesity, weight loss, and employment prospects evidence from a randomized trial, *Journal of Human Resources* (forthcoming).
- Roodman, D. (2011). Fitting fully observed recursive mixed-process models with cmp, *Stata Journal* **11**: 159–206.
- Sabia, J. J. and Rees, D. I. (2008). The effect of adolescent virginity status on psychological wellbeing, *Journal of Health Economics* **27**(5): 1368–1381.
- Sabia, J. J. and Rees, D. I. (2011). The effect of body weight on adolescent sexual activity, *Health Economics* **20**(11): 1330–1348.
- Schafer, M. H., Mustillo, S. A. and Ferraro, K. F. (2013). Age and the tenses of life satisfaction, *Journals of Gerontology, Series B: Psychological Sciences and Social Sciences* 68(4): 571–579.
- Taris, T. W. (2000). Quality of Mother-Child Interaction and the Intergenerational Transmission of Sexual Values: A Panel Study, *The Journal of Genetic Psychology* **161**(2): 169–181.

- Volpp, K. G., John, L. K., Troxel, A. B., Norton, L., Fassbender, J. and Loewenstein, G. (2008). Financial Incentive-Based Approaches for Weight Loss, *JAMA: The Journal of the American Medical Association* **300**(22): 2631–2637.
- Woloski-Wruble, A. C., Oliel, Y., Leefsma, M. and Hochner-Celnikier, D. (2010). Sexual Activities, Sexual and Life Satisfaction, and Successful Aging in Women, *The Journal of Sexual Medicine* 7(7): 2401–2410.
- Wooldridge, J. M. (2002). *Econometric Analysis of Cross Section and Panel Data*, MIT Press, Cambridge Massachusetts.
- World Health Organization (2000). Obesity: Preventing and managing the global epidemic, *Technical Report 894*, Geneva.

A Appendix

A.1 Results for 'living with a partner' (*together*₂₂) as dependent variable

	Mal	es	Females		All	
	Est. Coef.	S.E.	Est. Coef.	S.E.	Est. Coef.	S.E.
main equation (dependent variable: <i>together</i> ₂₂)						
ΔBMI_{1-10}	0.038	0.108	0.159	0.195	0.086	0.092
age	0.018	0.018	0.000	0.023	0.011	0.014
female	-	_	-	-	-0.212	0.237
together ₀	3.321**	0.356	3.109**	0.517	3.209**	0.288
BMI ₀	0.026	0.028	0.036	0.034	0.030	0.021
constant	-3.481^{**}	1.590	-2.830^{+}	1.928	-3.091^{**}	1.187
control function (dependent variable: ΔBMI_{1-10})						
incentive 150 ₁₋₄	-0.803^{*}	0.413	-0.715	0.519	-0.812^{**}	0.324
incentive 300 ₁₋₄	-1.008^{**}	0.418	-1.184^{**}	0.496	-1.110^{**}	0.321
incentive 250 ₅₋₁₀	-2.702^{**}	0.423	-1.397^{**}	0.495	-2.248^{**}	0.325
incentive 500 ₅₋₁₀	-2.036^{**}	0.409	-1.771^{**}	0.531	-1.886^{**}	0.326
age	0.004	0.021	-0.007	0.023	0.001	0.016
female	-	-	-	-	0.079	0.271
together ₀	-0.490	0.381	0.277	0.420	-0.170	0.290
BMI ₀	-0.082^{**}	0.028	0.012	0.029	-0.047^{**}	0.021
constant	3.579**	1.641	-0.151	1.643	2.081*	1.194
ancillary parameters						
σ (control function error S.D.)	2.654**	0.114	2.255**	0.134	2.556**	0.089
ρ (error correlation)	-0.063	0.333	-0.136	0.491	-0.146	0.266
# of observations (over all)	27	1	14	1	412	2
# of observations (main equation)	20	8	10	6	314	4
log likelihood	-69	1.2	-34	1.7	-104	1.1
instrument relevance (<i>p</i> -value, joint sig. instruments)	0.00	00	0.00	00	0.00	00
sample split (<i>p</i> -value, <i>LR</i> -test)	-		-		0.22	22

Table 11: Living with a Partner (estimated coefficients)

Notes: * significant at 5%; + significant at 10%.

	Mal	les	Females		All	
	Est. Coef.	S.E.	Est. Coef.	S.E.	Est. Coef.	S.E.
main equation (dependent variable: <i>together</i> ₂₂)						
ΔBMI_{1-22}	0.034	0.191	0.086	0.224	0.133	0.128
age	0.017	0.019	0.000	0.022	0.008	0.014
female	-	-	-	-	-0.229	0.237
together ₀	3.406**	0.575	3.094**	0.474	3.248**	0.412
BMI ₀	0.030	0.037	0.031	0.033	0.036*	0.022
constant	-3.636*	1.882	-2.806^{+}	1.823	-3.253**	1.173
control function (dependent variable: ΔBMI_{1-22})						
incentive 150 ₁₋₄	-0.660	0.624	-1.307^{*}	0.773	-0.914^{*}	0.477
incentive 300 ₁₋₄	-0.851	0.629	-1.598^{**}	0.761	-1.157^{**}	0.468
incentive 250 ₅₋₁₀	-1.790^{**}	0.625	-0.545	0.755	-1.275^{**}	0.479
incentive 500 ₅₋₁₀	-0.918^{+}	0.618	-1.494^{*}	0.806	-1.021^{**}	0.482
age	0.015	0.032	0.009	0.035	0.012	0.024
female	-	-	-	-	0.219	0.399
together ₀	-0.990^{*}	0.551	0.344	0.635	-0.446	0.425
BMI ₀	-0.123^{**}	0.042	-0.001	0.049	-0.080^{**}	0.032
constant	4.976**	2.449	0.388	2.556	3.257*	1.800
ancillary parameters						
σ (control function error S.D.)	3.416**	0.167	2.981**	0.204	3.318**	0.132
ρ (error correlation)	0.106	0.664	-0.031	0.711	-0.246	0.460
# of observations (over all)	20	9	10	7	316	5
# of observations (main equation)	20	7	10	6	313	3
log likelihood	-59	4.3	-29	5.8	-895	5.9
instrument relevance (<i>p</i> -value, joint sig. instruments)	0.00)6	0.05	52	0.00)1
sample split (<i>p</i> -value, <i>LR</i> -test)	-		-		0.56	6

Table 12: Living with a Partner, weight change over 22 months (est. coef.)

Notes: * significant at 5%; + significant at 10%.

A.2 Results for spec. considering weight change over 22 months (ΔBMI_{1-22})

	Ma	les	Fema	ales	All	
	Est. Coef.	S.E.	Est. Coef.	S.E.	Est. Coef.	S.E.
main equation (dependent variable: <i>sexpartner</i> ₁₁₋₂₂)						
ΔBMI_{1-22}	-0.233^{**}	0.061	-0.039	0.236	-0.207^{*}	0.114
age	0.010	0.016	-0.028	0.025	-0.005	0.015
female	-	_	-	_	-0.576^{**}	0.287
together ₀	0.798^{+}	0.506	1.642*	0.896	1.362**	0.463
BMI ₀	-0.057^{**}	0.024	0.045	0.044	-0.047^{*}	0.024
constant	1.575	1.211	-1.010	1.822	1.798^{+}	1.144
control function (dependent variable: ΔBMI_{1-22})						
incentive 150 ₁₋₄	-0.378	0.584	-1.563^{**}	0.723	-0.747	0.519
incentive 300 ₁₋₄	-0.726	0.565	-1.648^{**}	0.730	-1.044^{**}	0.492
incentive 250 ₅₋₁₀	-1.863^{**}	0.575	-0.435	0.679	-1.417^{**}	0.465
incentive 500 ₅₋₁₀	-1.051^{*}	0.544	-1.305^{+}	0.824	-1.023^{**}	0.474
age	0.015	0.032	0.010	0.035	0.012	0.024
female	-	_	-	-	0.207	0.399
together ₀	-0.974^{*}	0.551	0.312	0.635	-0.435	0.425
BMI ₀	-0.123^{**}	0.042	-0.002	0.049	-0.079^{**}	0.032
constant	4.860**	2.430	0.433	2.557	3.170*	1.804
ancillary parameters						
σ (control function error S.D.)	3.418**	0.167	2.984**	0.204	3.319**	0.132
ρ (error correlation)	0.793**	0.203	-0.688^{*}	0.370	0.456	0.443
# of observations (over all)	20	9	10	7	310	5
# of observations (main equation)	10	3	53	3	150	5
log likelihood	-58	8.3	-28	9.0	-88	9.6
instrument relevance (<i>p</i> -value, joint sig. instruments)	0.0	07	0.03	57	0.00)1
sample split (<i>p</i> -value, <i>LR</i> -test)	_	-	_		0.00)4

Table 13: Sexual Relationship, ΔBMI_{1-22} (estimated coefficients)

Notes: ** significant at 5%; * significant at 10%; + significant at 15%.

	Mal	es	Females		All	
	Est. Coef.	S.E.	Est. Coef.	S.E.	Est. Coef.	S.E.
main equation (dependent variable: <i>sexfrequency</i> ₁₁₋₂₂)						
ΔBMI_{1-22}	-0.037	0.149	0.143	0.121	-0.058	0.134
age	-0.037^{**}	0.019	-0.015	0.017	-0.025^{*}	0.013
female	-	-	-	-	-0.478^{**}	0.216
together ₀	0.661**	0.332	0.964^{+}	0.617	1.086**	0.228
BMI ₀	-0.056^{**}	0.028	0.010	0.027	-0.045^{*}	0.023
control function (dependent variable: ΔBMI_{1-22})						
incentive 150 ₁₋₄	-0.708	0.605	-1.794^{**}	0.672	-0.900^{*}	0.500
incentive 300 ₁₋₄	-0.869^{+}	0.595	-1.373^{*}	0.743	-1.136^{**}	0.483
incentive 250 ₅₋₁₀	-1.737^{**}	0.613	-0.330	0.561	-1.314^{**}	0.474
incentive 500 ₅₋₁₀	-1.000^{+}	0.610	-0.964	0.789	-1.000^{**}	0.503
age	0.015	0.032	0.011	0.035	0.012	0.024
female	-	-	-	-	0.218	0.400
together ₀	-0.994^{*}	0.551	0.247	0.638	-0.445	0.425
BMI ₀	-0.124^{**}	0.042	-0.007	0.049	-0.079^{**}	0.032
constant	5.069**	2.435	0.494	2.572	3.239*	1.800
ancillary parameters						
threshold 1	-4.680^{**}	1.407	-0.182	1.225	-3.235**	1.069
threshold 2	-3.673**	1.391	0.187	1.246	-2.396**	1.063
σ (control function error S.D.)	3.416**	0.167	3.005**	0.208	3.318**	0.132
ho (error correlation)	0.201	0.518	-0.880^{**}	0.143	0.040	0.460
# of observations (over all)	209)	10	7	316	5
# of observations (main equation)	99		53		152	2
log likelihood	-632	2.5	-30	4.4	-954	4.1
instrument relevance (<i>p</i> -value, joint sig. instruments)	0.00)6	0.10)8	0.00	1
sample split (p-value, LR-test)	_		-		0.00	1

Table 14: Frequency of Sexual Intercourse	ΔBMI_{1-22}	(estimated	coefficients)
---	---------------------	------------	---------------

Notes: ** significant at 5%; * significant at 10%; + significant at 15%.

A.3 Results for linear 2SLS estimation

A.3.1 Basic Model

•		1			
Mal	es	Fema	iles	Al	1
Est. Coef.	S.E.	Est. Coef.	S.E.	Est. Coef.	S.E.
sexpartner ₁₁₋₂₂)					
-0.039^{+}	0.025	-0.012	0.061	-0.035^{+}	0.024
0.001	0.004	-0.008	0.009	-0.002	0.004
-	-	-	-	-0.182^{**}	0.072
0.437**	0.105	0.577**	0.131	0.479**	0.077
-0.012^{+}	0.008	-0.006	0.014	-0.011^{+}	0.007
0.822**	0.418	0.806	0.654	0.890**	0.346
ble: ΔBMI_{1-10})					
-0.796^{**}	0.368	-0.727^{+}	0.503	-0.808^{**}	0.289
-0.997^{**}	0.403	-1.196^{**}	0.462	-1.103^{**}	0.298
-2.714^{**}	0.456	-1.382^{**}	0.502	-2.260**	0.353
-2.031^{**}	0.401	-1.772^{**}	0.541	-1.877^{**}	0.310
0.004	0.021	-0.007	0.030	0.001	0.018
-	-	-	-	0.078	0.273
-0.490	0.416	0.275	0.435	-0.170	0.314
-0.082^{*}	0.046	0.012	0.039	-0.046^{+}	0.032
3.566^{+}	2.217	-0.152	2.316	2.075	1.570
27	1	14	1	412	2
103	3	53	5	156	6
17.8	12	5.31	13	22.1	21
	$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	$\begin{tabular}{ c c c c c } \hline Males & \hline Est. Coef. & S.E. \\ \hline \hline Est. Coef. & S.E. \\ \hline \hline expartner_{11-22}) & & & & \\ \hline -0.039^+ & 0.025 & & \\ 0.001 & 0.004 & & & \\ \hline 0.437^{**} & 0.105 & & & \\ \hline 0.437^{**} & 0.105 & & & \\ \hline 0.437^{**} & 0.105 & & & \\ \hline 0.012^+ & 0.008 & & & \\ 0.822^{**} & 0.418 & & & \\ \hline 0.822^{**} & 0.418 & & & \\ \hline 0.822^{**} & 0.403 & & & \\ \hline -0.796^{**} & 0.368 & & & \\ \hline 0.997^{**} & 0.403 & & & \\ \hline -0.796^{**} & 0.403 & & & \\ \hline -2.714^{**} & 0.456 & & & \\ \hline -2.031^{**} & 0.401 & & \\ 0.004 & 0.021 & & & \\ \hline -0.490 & 0.416 & & & \\ \hline -0.082^* & 0.046 & & \\ \hline 3.566^+ & 2.217 & & \\ \hline 271 & & & \\ 103 & & & \\ 103 & & & \\ 17.812 & & & \\ \hline \end{tabular}$	$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	$\begin{tabular}{ c c c c c c } \hline Males & Females \\ \hline \hline Bst. Coef. & S.E. & Est. Coef. & S.E. \\ \hline \hline Est. Coef. & S.E. & Est. Coef. & S.E. \\ \hline \hline expartmer_{11-22}) & & & & & \\ \hline -0.039^+ & 0.025 & -0.012 & 0.061 \\ 0.001 & 0.004 & -0.008 & 0.009 \\ \hline \hline 0.437^{**} & 0.105 & 0.577^{**} & 0.131 \\ -0.012^+ & 0.008 & -0.006 & 0.014 \\ 0.822^{**} & 0.418 & 0.806 & 0.654 \\ \hline \hline ble: ΔBMI_{1-10}) & & & & \\ \hline -0.796^{**} & 0.368 & -0.727^+ & 0.503 \\ -0.997^{**} & 0.403 & -1.196^{**} & 0.462 \\ -2.714^{**} & 0.456 & -1.382^{**} & 0.502 \\ -2.031^{**} & 0.401 & -1.772^{**} & 0.541 \\ 0.004 & 0.021 & -0.007 & 0.030 \\ \hline - & & & & & & \\ \hline - & & & & & & & \\ \hline 271 & & 141 \\ 103 & & & 53 \\ 17.812 & & & 5.313 \\ \hline \end{tabular}$	$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$

Table 15: 2SLS, Sexual Relationship (estimated coefficients)

Notes: ** significant at 5%; * significant at 10%; + significant at 15%; bootstrapped standard errors reported.

Table 16: 2SLS, Fre	quency of Sexual	l Intercourse (es	stimated coefficients)
---------------------	------------------	-------------------	------------------------

	Males		Females		All	
	Est. Coef.	S.E.	Est. Coef.	S.E.	Est. Coef.	S.E.
main equation (dependent variable: s	exfrequency ₁₁₋₂₂)					
ΔBMI_{1-10}	-0.020	0.048	0.015	0.116	-0.027	0.045
age	-0.018^{**}	0.007	-0.014	0.017	-0.013^{+}	0.008
female	-	-	-	-	-0.303^{**}	0.132
together ₀	0.409**	0.171	1.096**	0.239	0.686**	0.140
BMI0	-0.033^{**}	0.015	-0.016	0.024	-0.027^{**}	0.012
constant	3.225**	0.678	1.720^{+}	1.170	2.546**	0.617
first stage equation (dependent varial	ole: ΔBMI_{1-10})					
incentive 150 ₁₋₄	-0.796^{**}	0.370	-0.727^{+}	0.489	-0.808^{**}	0.314
incentive 300 ₁₋₄	-0.997^{**}	0.395	-1.196^{**}	0.472	-1.103^{**}	0.305
incentive 250 ₅₋₁₀	-2.714^{**}	0.455	-1.382^{**}	0.506	-2.260^{**}	0.362
incentive 500 ₅₋₁₀	-2.031^{**}	0.392	-1.772^{**}	0.545	-1.877^{**}	0.313
age	0.004	0.021	-0.007	0.031	0.001	0.018
female	-	-	-	-	0.078	0.279
together ₀	-0.490	0.412	0.275	0.436	-0.170	0.315
BMI0	-0.082^{*}	0.045	0.012	0.040	-0.046	0.033
constant	3.566*	2.166	-0.152	2.344	2.075	1.605
# of observations (first stage)	271		141		412	
# of observations (main equation)	99)	53		152	
instrument relevance (F-statistic)	17.812		5.313		22.121	

Notes: ** significant at 5%; * significant at 10%; + significant at 15%; bootstrapped standard errors reported.

A.3.2 Conditional Model

	Males		Females		All	
	Est. Coef.	S.E.	Est. Coef.	S.E.	Est. Coef.	S.E.
main equation (dependent variable: s	sexpartner ₁₁₋₂₂)					
ΔBMI_{5-10}	-0.149^{+}	0.097	0.315	0.554	-0.082	0.107
age	-0.003	0.007	0.012	0.038	-0.005	0.006
female	-	_	-	-	-0.098	0.096
together ₄	0.341**	0.139	0.273	0.534	0.422**	0.106
BMI4	-0.017	0.013	-0.046	0.047	-0.018^{+}	0.011
constant	1.318**	0.600	1.381	1.559	1.370**	0.451
first stage equation (dependent varial	ble: ΔBMI_{5-10})					
incentive 250 ₅₋₁₀	-1.417^{**}	0.407	-0.366	0.520	-1.034^{**}	0.323
incentive 500 ₅₋₁₀	-0.992^{**}	0.405	-0.416	0.656	-0.738^{**}	0.339
age	0.004	0.019	-0.049	0.036	-0.014	0.017
female	_	_	_	_	0.116	0.295
together ₄	-0.102	0.414	0.822	0.577	0.265	0.339
BMI4	-0.030	0.041	0.080^{+}	0.050	0.012	0.032
constant	2.073	1.585	-0.245	2.570	1.012	1.361
# of observations (over all)	179		81		260	
# of observations (main equation)	65	;	32		97	
instrument relevance (F-statistic)	6.442		0.299		5.127	

Table 17: 2SLS, Sexual Relationship, cond. on success in weight-loss phase (est. coefficients)

Notes: ** significant at 5%; * significant at 10%; + significant at 15%; bootstrapped standard errors reported. All model equations estimated using only individuals who successfully lost body weight during month one to four.

T-1-1-10. OCIC E.	new of Internetion	and an average f	in markelet lage.	where (ast is affinized to)
1able 16: 25L5, FI	rea. of intercource.	cond. on success I	in weight-loss	Dhase (est. coefficients)

	Males		Females		All	
	Est. Coef.	S.E.	Est. Coef.	S.E.	Est. Coef.	S.E.
main equation (dependent variable: s	exfrequency ₁₁₋₂₂)					
ΔBMI_{5-10}	-0.219	0.203	0.438	1.214	-0.163	0.233
age	-0.019^{*}	0.011	0.010	0.071	-0.017	0.012
female	-	-	-	-	-0.094	0.181
$together_4$	0.444^{*}	0.247	0.763	1.169	0.735**	0.212
BMI_4	-0.040^{*}	0.023	-0.080	0.102	-0.038^{*}	0.020
constant	3.481**	0.950	2.880	3.329	3.101**	0.860
first stage equation (dependent varial	ble: ΔBMI_{5-10})					
incentive 2505-10	-1.417^{**}	0.397	-0.366	0.535	-1.034^{**}	0.322
incentive 500 ₅₋₁₀	-0.992^{**}	0.412	-0.416	0.653	-0.738^{**}	0.336
age	0.004	0.019	-0.049	0.037	-0.014	0.018
female	-	-	-	-	0.116	0.294
together ₄	-0.102	0.407	0.822	0.582	0.265	0.353
BMI4	-0.030	0.041	0.080^{+}	0.050	0.012	0.033
constant	2.073	1.570	-0.245	2.645	1.012	1.387
# of observations (over all)	179		81		260	
# of observations (main equation)	63	5	32		95	
instrument relevance (F-statistic)	6.442		0.299		5.127	

Notes: ** significant at 5%; * significant at 10%; + significant at 15%; bootstrapped standard errors reported. All model equations estimated using only individuals who successfully lost body weight during month one to four.

Diskussionspapiere 2015 Discussion Papers 2015

- 01/2015 **Seebauer, Michael:** Does Direct Democracy Foster Efficient Policies? An Experimental Investigation of Costly Initiatives
- 02/2015 **Bünnings, Christian, Schmitz, Hendrik, Tauchmann, Harald and Ziebarth, Nicolas R.**: How Health Plan Enrollees Value Prices Relative to Supplemental Benefits and Service Quality
- 03/2015 **Schnabel, Claus:** United, yet apart? A note on persistent labour market differences between western and eastern Germany
- 04/2015 Dieckmann, Anja, Fischbacher, Urs, Grimm, Veronika, Unfried, Matthias, Utikal, Verena and Valmasoni, Lorenzo: Trust and Beliefs among Europeans: Cross-Country Evidence on Perceptions and Behavior
- 05/2015 **Grimm, Veronika, Utikal, Verena and Valmasoni, Lorenzo**: In-group favoritism and discrimination among multiple out-groups

Diskussionspapiere 2014 Discussion Papers 2014

- 01/2014 **Rybizki, Lydia**: Learning cost sensitive binary classification rules accounting for uncertain and unequal misclassification costs
- 02/2014 Abbiati, Lorenzo, Antinyan, Armenak and Corazzini, Lucca: Are Taxes Beautiful? A Survey Experiment on Information, Tax Choice and Perceived Adequacy of the Tax Burden
- 03/2014 **Feicht, Robert, Grimm, Veronika and Seebauer, Michael**: An Experimental Study of Corporate Social Responsibility through Charitable Giving in Bertrand Markets
- 04/2014 Grimm, Veronika, Martin, Alexander, Weibelzahl, Martin and Zoettl, Gregor: Transmission and Generation Investment in Electricity Markets: The Effects of Market Splitting and Network Fee Regimes
- 05/2014 **Cygan-Rehm, Kamila and Riphahn, Regina**: Teenage Pregnancies and Births in Germany: Patterns and Developments
- 06/2014 Martin, Alexander and Weibelzahl, Martin: Where and when to Pray? -Optimal Mass Planning and Efficient Resource Allocation in the Church
- 07/2014 Abraham, Martin, Lorek, Kerstin, Richter, Friedemann and Wrede, Matthias: Strictness of Tax Compliance Norms: A Factorial Survey on the Acceptance of Inheritance Tax Evasion in Germany

- 08/2014 Hirsch, Boris, Oberfichtner, Michael and Schnabel Claus: The levelling effect of product market competition on gender wage discrimination
- 09/1014 Mangold, Benedikt: Plausible Prior Estimation
- 10/2014 Gehrke, Britta: Fiscal Rules and Unemployment
- 11/2014 **Gehrke, Britta** and **Yao, Fang**: Phillips Curve Shocks and Real Exchange Rate Fluctuations: SVAR Evidence
- 12/2014 Mäder, Miriam, Müller, Steffen, Riphahn, Regina and Schwientek, Caroline: Intergenerational transmission of unemployment - evidence for German sons
- 13/2014 **Casal, Sandro, Ploner, Matteo and Sproten, Alec N.**: Fostering the Best Execution Regime. An Experiment about Pecuniary Sanctions and Accountability in Fiduciary Money Management
- 14/2014 **Lochner, Benjamin**: Employment protection in dual labor markets Any amplification of macroeconomic shocks?
- 15/2014 Herrmann, Klaus, Teis, Stefan and Yu, Weijun: Components of Intraday Volatility and Their Prediction at Different Sampling Frequencies with Application to DAX and BUND Futures

Diskussionspapiere 2013 Discussion Papers 2013

- 01/2013 Wrede, Matthias: Rational choice of itemized deductions
- 02/2013 Wrede, Matthias: Fair Inheritance Taxation in the Presence of Tax Planning
- 03/2013 **Tinkl, Fabian:** Quasi-maximum likelihood estimation in generalized polynomial autoregressive conditional heteroscedasticity models
- 04/2013 **Cygan-Rehm, Kamila**: Do Immigrants Follow Their Home Country's Fertility Norms?
- 05/2013 Ardelean, Vlad and Pleier, Thomas: Outliers & Predicting Time Series: A comparative study
- 06/2013 **Fackler, Daniel and Schnabel, Claus**: Survival of spinoffs and other startups: First evidence for the private sector in Germany, 1976-2008
- 07/2013 **Schild, Christopher-Johannes:** Do Female Mayors Make a Difference? Evidence from Bavaria

- 08/2013 **Brenzel, Hanna, Gartner, Hermann and Schnabel Claus**: Wage posting or wage bargaining? Evidence from the employers' side
- 09/2013 Lechmann, Daniel S. and Schnabel Claus: Absence from work of the selfemployed: A comparison with paid employees
- 10/2013 **Bünnings, Ch. and Tauchmann, H**.: Who Opts Out of the Statutory Health Insurance? A Discrete Time Hazard Model for Germany

Diskussionspapiere 2012 Discussion Papers 2012

- 01/2012 Wrede, Matthias: Wages, Rents, Unemployment, and the Quality of Life
- 02/2012 Schild, Christopher-Johannes: Trust and Innovation Activity in European Regions - A Geographic Instrumental Variables Approach
- 03/2012 **Fischer, Matthias:** A skew and leptokurtic distribution with polynomial tails and characterizing functions in closed form
- 04/2012 Wrede, Matthias: Heterogeneous Skills and Homogeneous Land: Segmentation and Agglomeration
- 05/2012 Ardelean, Vlad: Detecting Outliers in Time Series

Diskussionspapiere 2011 Discussion Papers 2011

- 01/2011 Klein, Ingo, Fischer, Matthias and Pleier, Thomas: Weighted Power Mean Copulas: Theory and Application
- 02/2011 **Kiss, David:** The Impact of Peer Ability and Heterogeneity on Student Achievement: Evidence from a Natural Experiment
- 03/2011 **Zibrowius, Michael:** Convergence or divergence? Immigrant wage assimilation patterns in Germany
- 04/2011 Klein, Ingo and Christa, Florian: Families of Copulas closed under the Construction of Generalized Linear Means
- 05/2011 **Schnitzlein, Daniel**: How important is the family? Evidence from sibling correlations in permanent earnings in the US, Germany and Denmark

- 06/2011 **Schnitzlein, Daniel**: How important is cultural background for the level of intergenerational mobility?
- 07/2011 **Steffen Mueller:** Teacher Experience and the Class Size Effect Experimental Evidence
- 08/2011 Klein, Ingo: Van Zwet Ordering for Fechner Asymmetry
- 09/2011 **Tinkl, Fabian and Reichert Katja:** Dynamic copula-based Markov chains at work: Theory, testing and performance in modeling daily stock returns
- 10/2011 **Hirsch, Boris and Schnabel, Claus:** Let's Take Bargaining Models Seriously: The Decline in Union Power in Germany, 1992 – 2009
- 11/2011 **Lechmann, Daniel S.J. and Schnabel**, **Claus** : Are the self-employed really jacks-of-all-trades? Testing the assumptions and implications of Lazear's theory of entrepreneurship with German data
- 12/2011 Wrede, Matthias: Unemployment, Commuting, and Search Intensity
- 13/2011 Klein, Ingo: Van Zwet Ordering and the Ferreira-Steel Family of Skewed Distributions

Diskussionspapiere 2010 Discussion Papers 2010

- 01/2010 **Mosthaf, Alexander, Schnabel, Claus and Stephani, Jens:** Low-wage careers: Are there dead-end firms and dead-end jobs?
- 02/2010 Schlüter, Stephan and Matt Davison: Pricing an European Gas Storage Facility using a Continuous-Time Spot Price Model with GARCH Diffusion
- 03/2010 **Fischer, Matthias, Gao, Yang and Herrmann, Klaus:** Volatility Models with Innovations from New Maximum Entropy Densities at Work
- 04/2010 Schlüter, Stephan and Deuschle, Carola: Using Wavelets for Time Series Forecasting – Does it Pay Off?
- 05/2010 Feicht, Robert and Stummer, Wolfgang: Complete closed-form solution to a stochastic growth model and corresponding speed of economic recovery.
- 06/2010 **Hirsch, Boris and Schnabel, Claus:** Women Move Differently: Job Separations and Gender.

07/2010 **Gartner, Hermann, Schank, Thorsten and Schnabel, Claus:** Wage cyclicality under different regimes of industrial relations.

08/2010 **Tinkl, Fabian**: A note on Hadamard differentiability and differentiability in quadratic mean.

Diskussionspapiere 2009 Discussion Papers 2009

- 01/2009 Addison, John T. and Claus Schnabel: Worker Directors: A German Product that Didn't Export?
- 02/2009 **Uhde, André and Ulrich Heimeshoff:** Consolidation in banking and financial stability in Europe: Empirical evidence
- 03/2009 **Gu, Yiquan and Tobias Wenzel:** Product Variety, Price Elasticity of Demand and Fixed Cost in Spatial Models
- 04/2009 Schlüter, Stephan: A Two-Factor Model for Electricity Prices with Dynamic Volatility
- 05/2009 Schlüter, Stephan and Fischer, Matthias: A Tail Quantile Approximation Formula for the Student t and the Symmetric Generalized Hyperbolic Distribution
- 06/2009 Ardelean, Vlad: The impacts of outliers on different estimators for GARCH processes: an empirical study
- 07/2009 Herrmann, Klaus: Non-Extensitivity versus Informative Moments for Financial Models - A Unifying Framework and Empirical Results
- 08/2009 Herr, Annika: Product differentiation and welfare in a mixed duopoly with regulated prices: The case of a public and a private hospital
- 09/2009 **Dewenter, Ralf, Haucap, Justus and Wenzel, Tobias**: Indirect Network Effects with Two Salop Circles: The Example of the Music Industry
- 10/2009 **Stuehmeier, Torben and Wenzel, Tobias**: Getting Beer During Commercials: Adverse Effects of Ad-Avoidance
- 11/2009 Klein, Ingo, Köck, Christian and Tinkl, Fabian: Spatial-serial dependency in multivariate GARCH models and dynamic copulas: A simulation study
- 12/2009 Schlüter, Stephan: Constructing a Quasilinear Moving Average Using the Scaling Function

- 13/2009 Blien, Uwe, Dauth, Wolfgang, Schank, Thorsten and Schnabel, Claus: The institutional context of an "empirical law": The wage curve under different regimes of collective bargaining
- 14/2009 Mosthaf, Alexander, Schank, Thorsten and Schnabel, Claus: Low-wage employment versus unemployment: Which one provides better prospects for women?

Diskussionspapiere 2008 Discussion Papers 2008

- 01/2008 Grimm, Veronika and Gregor Zoettl: Strategic Capacity Choice under Uncertainty: The Impact of Market Structure on Investment and Welfare
- 02/2008 **Grimm, Veronika and Gregor Zoettl:** Production under Uncertainty: A Characterization of Welfare Enhancing and Optimal Price Caps
- 03/2008 **Engelmann, Dirk and Veronika Grimm**: Mechanisms for Efficient Voting with Private Information about Preferences
- 04/2008 Schnabel, Claus and Joachim Wagner: The Aging of the Unions in West Germany, 1980-2006
- 05/2008 Wenzel, Tobias: On the Incentives to Form Strategic Coalitions in ATM Markets
- 06/2008 Herrmann, Klaus: Models for Time-varying Moments Using Maximum Entropy Applied to a Generalized Measure of Volatility
- 07/2008 Klein, Ingo and Michael Grottke: On J.M. Keynes' "The Principal Averages and the Laws of Error which Lead to Them" - Refinement and Generalisation